IOWA STATE UNIVERSITY Digital Repository

Graduate Theses and Dissertations

Graduate College

2014

Quantifying the economic competitiveness of cellulosic biofuel pathways under uncertainty and regional sensitivity

Tristan Brown Iowa State University

Follow this and additional works at: http://lib.dr.iastate.edu/etd Part of the <u>Engineering Commons</u>, <u>Finance and Financial Management Commons</u>, and the <u>Oil,</u> <u>Gas, and Energy Commons</u>

Recommended Citation

Brown, Tristan, "Quantifying the economic competitiveness of cellulosic biofuel pathways under uncertainty and regional sensitivity" (2014). *Graduate Theses and Dissertations*. 13884. http://lib.dr.iastate.edu/etd/13884

This Dissertation is brought to you for free and open access by the Graduate College at Iowa State University Digital Repository. It has been accepted for inclusion in Graduate Theses and Dissertations by an authorized administrator of Iowa State University Digital Repository. For more information, please contact digirep@iastate.edu.

Quantifying the economic competitiveness of cellulosic biofuel pathways under uncertainty and regional sensitivity

by

Tristan R. Brown

A dissertation submitted to the graduate faculty

in partial fulfillment of the requirements for the degree of

DOCTOR OF PHILOSOPHY

Major: Biorenewable Resources and Technology

Program of Study Committee: Mark Mba-Wright, Major Professor Jacqulyn Baughman Guiping Hu Lulu Rodriguez Amy Erica Smith

Iowa State University

Ames, Iowa

TABLE OF CONTENTS

LIST OF	F FIGURES	iv
LIST OF	F TABLES	vi
ACKNC	DWLEDGEMENTS	viii
ABSTR	ACT	ix
CHAPT	ER 1. INTRODUCTION	1
1.1.	Background	1
1.2.	Dissertation Objectives	
1.3.	Intellectual Merit	
1.4.	Dissertation Organization	
CHAPT	ER 2. TECHNO-ECONOMIC IMPACTS OF SHALE GAS ON CELLUL	OSIC
BIOFUE	EL PATHWAYS	
2.1.	Abstract	
2.2.	Introduction	24
2.3.	Methodology	
2.4.	Results and Discussion	
2.4.	1. Uncertainty Analysis Based on 2010 Projections	
2.4.	2. Uncertainty Analysis Based on 2013 Projections	
2.5.	Conclusion	
2.6.	Supporting Materials	
2.6.	1. Cellulosic Biofuel Pathway Descriptions	
2.6.	2. Fast pyrolysis and Hydroprocessing	
2.6.	3. Gasification and AAS	
2.6.	4. Gasification and FTS	
2.6.	5. Gasification and MTG Synthesis	
2.6.	6. Enzymatic Hydrolysis and Fermentation	
CHAPT	ER 3. A FRAMEWORK FOR DEFINING THE ECONOMIC FEASIBILI	TY OF
CELLU	LOSIC BIOFUEL PATHWAYS	
3.1.	Abstract	
3.2.	Introduction	
3.3.	Methodology	
3.3.	1. Cost of equity	61
3.3.	2. Cost of debt	
3.3.	3. Monte Carlo simulation	67
3.4.	Results	69
3.5.	Sensitivity analysis	72
3.6.	Conclusion	75
3.7.	Acknowledgements	76

CHAPT	ER 4	A. QUANTIFYING THE ECONOMIC COMPETITIVENESS OF INITIAL	
U.S. CE	ELLU	LOSIC BIOFUEL PATHWAYS	. 77
4.1.	Abs	stract	. 77
4.2.	Intr	oduction	. 77
4.2.1.	Me	thodology	. 83
4.2	.2.	Scenario selection	. 83
4.2	.3.	Feedstock cost	. 86
4.2	.4.	Location capital cost factor	. 89
4.2	.5.	State corporate income tax rate	. 90
4.2	.6.	Energy commodity prices	. 90
4.2	.7.	Uncertainty analysis	. 91
4.3.	Res	ults	. 96
4.4.	Cor	nclusion	. 98
СНАРТ	ER 5	5. CONCLUSIONS AND RECOMMENDATIONS FOR FUTURE WORK	. 99
5.1.	Cor	nclusions	. 99
5.2.	Rec	commendations for Future Work	101
REFER	ENC	ES	105

LIST OF FIGURES

Figure 1.1.	RFS1 blending mandates versus actual production (1,2)1
Figure 1.2.	Total blending volumes under the RFS1 and RFS2, 2008-2012 (2)2
Figure 1.3.	Annual volumetric mandates for each category under the RFS2 (2). *The biomass-based diesel category is set annually by the EPA, but the volume is not to fall below 1 billion gallons
Figure 1.4.	EPA adjustments to the cellulosic biofuels volumetric mandates, 2010-2014 (2). Actual volume for 2014 is not known at time of writing. Actual volumes for 2012 and 2013 were 0.02 million and 0.82 million gallons, respectively
Figure 1.5.	The E10 blend wall relative to the RFS2 volumes that do not exclude fuel ethanol (2,19)12
Figure 1.6.	RIN prices since May 2013 (22). D4 = Biomass-based diesel category; D5 = Advanced biofuel category; D6 = Total renewable fuel category13
Figure 2.1.	Monthly U.S. natural gas wellhead and WTI contract nominal price, 1983-2012 (56,57)25
Figure 2.2.	Historical (1997-2012) and projected (2013-2040) increases in annual nominal natural gas (\$/MMBTU) and West Texas Intermediate (WTI) petroleum (\$/bbl) spot prices (19)
Figure 2.3.	NPV comparison of biorefinery scenario uncertainty analysis based on 2010/2013 fuel price projections. Light shading indicates the AEO 2010 scenario and dark shading indicates the AEO 2013 scenario
Figure S2.1.	Methodology for the profitability analysis of biofuel scenario pathways based on commodity prices
Figure S2.2.	Comparison of price histograms and probability density function fits44
Figure S2.3.	Projected annual prices from AEO 2010 and AEO 2013
Figure S2.4.	Schematics of pathways analyzed. a – fast pyrolysis and hydroprocessing; b – gasification and AAS; c – gasification and FTS; d – gasification and MTG synthesis; e – enzymatic hydrolysis and fermentation
Figure 3.1.	Originally-mandated, revised, and actual RFS2 cellulosic biofuel blending volumes, 2010-2014. 2013 actual volume estimated based on blending data through October 2013 (2,5,24)

LIST OF TABLES

Table 1.1.	RFS2 biofuel categories and definitions (2).	3
Table 1.2.	RINs created per type of biofuel (5)	7
Table 1.3.	2014 volumetric mandates for the RFS2, both original and proposed (24).	14
Table 2.1.	Probability distribution function selection and expected value for each commodity price (79)	30
Table S2.1.	Process scenario mass balances and annual operating hours. All scenarios assume 2000 metric tons per day (MTPD) of biomass input and \$83 per MT feedstock cost. (MGY – million gallons per year)	42
Table S2.2.	Capital and operating costs of process scenarios adjusted 2011 US dollar cost basis	43
Table S2.3.	Compositions of biomass feedstocks	45
Table 3.1.	Pathway scenario mass balances and annual operating hours. All scenarios assume 2000 MT (dry)/day of biomass input.	58
Table 3.2.	Capital and operating costs of pathway scenarios adjusted 2011 US dollar cost basis	59
Table 3.3.	Probability distribution function for monthly change of each commodity price	61
Table 3.4.	Advanced biorenewable companies historical market betas	63
Table 3.5.	Publicly-traded independent corn ethanol companies	64
Table 3.6.	ICRs and ratings for low market-capitalization firms (117)	66
Table 3.7.	Pathway scenario interest coverage ratios and interest rates	67
Table 3.8.	Sensitivity of ICR to commodity prices	72
Table 3.9.	Sensitivity of results to financial assumptions	74
Table 4.1.	Location scenario details	85
Table 4.2.	Location scenario market conditions. Price ranges based on projections from 2013 to 2032	88
Table 4.3.	Probability distribution function for monthly change of each commodity price	93

Table 4.4.	Cost of debt results for each location scenario	95
Table 4.5.	Location scenario mean 20-year NPVs and 10-year probabilities of default	96

ACKNOWLEDGEMENTS

I would like to thank my Ph.D. advisor, Dr. Mark Mba-Wright, for providing intellectual and financial support to my research during my time as a Ph.D. student. My graduate school experience has been very unconventional and I am grateful to him for enabling me to pursue course development and teaching activities despite the frequent distractions that they have provided from my research. I would have been unable to reach this point in my career without his constant guidance in everything from implementing new research methodologies to the peer review process to surviving job interviews in academia. I would also like to thank my committee members, Drs. Jacqulyn Baughman, Guiping Hu, Lulu Rodriguez, and Amy Erica Smith both for agreeing to serve on a committee consisting of a wide diversity of disciplines and for providing my research with direction and meaning throughout my studies.

I would like to thank Jill Euken for providing me with numerous speaking and outreach opportunities during my time at Iowa State University. I would like to thank all of the staff at the Bioeconomy Institute for patiently guiding me through the paperwork that is required of direct admit students in interdisciplinary programs. I would also like to thank all of the students who have enrolled in BRT/POL S 515 and 516 over the years for demonstrating to me just how fulfilling of an experience teaching can be.

Finally, I would like to thank my parents and family for providing me with encouragement, advice, and welcome distractions, both before and during my time at Iowa State. I am above all grateful to my wife, Kate, for being an unflagging source of motivation, optimism, and inspiration throughout this experience while simultaneously maintaining her own strenuous academic schedule. I am truly fortunate.

viii

ABSTRACT

The revised Renewable Fuel Standard requires the annual blending of 16 billion gallons of cellulosic biofuel by 2022 from zero gallons in 2009. The necessary capacity investments have been underwhelming to date, however, and little is known about the likely composition of the future cellulosic biofuel industry as a result. This dissertation develops a framework for identifying and analyzing the industry's likely future composition while also providing a possible explanation for why investment in cellulosic biofuels capacity has been low to date.

The results of this dissertation indicate that few cellulosic biofuel pathways will be economically competitive with petroleum on an unsubsidized basis. Of five cellulosic biofuel pathways considered under 20-year price forecasts with volatility, only two achieve positive mean 20-year net present value (NPV) probabilities. Furthermore, recent exploitation of U.S. shale gas reserves and the subsequent fall in U.S. natural gas prices have negatively impacted the economic competitiveness of all but two of the cellulosic biofuel pathways considered; only two of the five pathways achieve substantially higher 20-year NPVs under a post-shale gas economic scenario relative to a pre-shale gas scenario.

The economic competitiveness of cellulosic biofuel pathways with petroleum is reduced further when considered under price uncertainty in combination with realistic financial assumptions. This dissertation calculates pathway-specific costs of capital for five cellulosic biofuel pathway scenarios. The analysis finds that the large majority of the scenarios incur costs of capital that are substantially higher than those commonly assumed in the literature. Employment of these costs of capital in a comparative TEA greatly reduces the

ix

mean 20-year NPVs for each pathway while increasing their 10-year probabilities of default to above 80% for all five scenarios.

Finally, this dissertation quantifies the economic competitiveness of six cellulosic biofuel pathways being commercialized in eight different U.S. states under price uncertainty, utilization of pathway-specific costs of capital, and region-specific economic factors. 10-year probabilities of default in excess of 60% are calculated for all eight location scenarios considered, with default probabilities in excess of 98% calculated for seven of the eight. Negative mean 20-year NPVs are calculated for seven of the eight location scenarios.

CHAPTER 1. INTRODUCTION

1.1. Background

In 2005 the U.S. Congress passed the Energy Policy Act of 2005, which created a national volumetric mandate for the blending of biofuel with conventional fuels. The mandate, named the Renewable Fuel Standard (RFS1), required escalating volumes of biofuel (see Figure 1.1), which at the time consisted almost entirely of corn ethanol, to be blended with gasoline prior to retail. Actual production exceeded the mandated volume by 23% in the first year of the RFS1, however, due in large part to the simultaneous but separate provision of a large refundable tax credit to corn ethanol producers in the form of the Volumetric Ethanol Excise Tax Credit (VEETC), commonly known as the "blender's credit."



Figure 1.1. RFS1 blending mandates versus actual production (1,2).

In late 2007 Congress passed the Energy Independence and Security Act of 2007. Among other things, the new legislation replaced the RFS1 with an expanded mandate known as the revised Renewable Fuel Standard (RFS2). The RFS2 is significantly larger than the RFS1 in terms of both its overall volumes and its complexity. The total biofuel blending volumes under the RFS2 between 2008 and 2012 are as much as 90% higher than for the same years under the RFS1, reflecting the ease with which the corn ethanol industry exceeded the latter's volumetric mandates (see Figure 1.2).



Figure 1.2. Total blending volumes under the RFS1 and RFS2, 2008-2012 (2).

The RFS2 also replaces the RFS1's single biofuel category with four categories, three of which encompass next-generation biofuel pathways. Furthermore, each biofuel category is strictly defined according to both feedstock and lifecycle greenhouse gas (GHG) emissions relative to conventional fuels (see Table 1.1). Finally, the various categories are each nested within the next-largest category, allowing a biofuel meeting the definition of a nested

category to also meet the definition of the larger category in which the nested category is located. The primary category, named "Total Renewable Fuels", encompasses all of the other categories and includes any biofuel produced from biomass feedstock that achieves a 20% GHG emissions reduction threshold relative to conventional fuel. This category's volumetric mandate grows from 9 billion gallons in 2008 to 36 billion gallons in 2022. Corn ethanol's contribution to the total renewable fuels category, regardless of its GHG emissions reduction threshold, is capped at 15 billion gallons per year from 2015 on due to prevailing concerns at the time of the mandate's creation that the use of corn-based biofuels in the U.S. drives chronic hunger (3) and rainforest destruction (4) in the developing world.

Table 1.1. RFS2 biofuel categories and definitions (2).

Mandate category	Feedstock	GHG reduction threshold
Total renewable fuels	Biomass (inc. corn starch)	20%
Advanced biofuels	Biomass (exc. corn starch)	50%
Biomass-based diesel	Non-lignocellulosic biomass (exc. corn starch)	50%
Cellulosic biofuels	Lignocellulosic biomass	60%

Residing within the total renewable fuels category is the "Advanced Biofuels" category. This second category encompasses all biofuels derived from biomass sources other than corn starch that achieve a 50% GHG emissions reduction threshold relative to conventional fuels. While the category's volume is non-existent in the first year of the RFS2, it increases to 21 billion gallons by 2022 (see Figure 1.3), making it larger after 2019 than the maximum corn starch-derived volume permitted by the mandate.



Figure 1.3. Annual volumetric mandates for each category under the RFS2 (2). *The biomass-based diesel category is set annually by the EPA, but the volume is not to fall below 1 billion gallons.

The advanced biofuels category encompasses the final two categories, which are the "Biomass-Based Diesel" and "Cellulosic Biofuels" categories. The biomass-based diesel category covers diesel and diesel-like biofuels (e.g., both renewable diesel and FAME biodiesel) that are derived from non-lignocellulosic biomass and achieve a 50% GHG emissions reduction. In 2013 approximately 86% of the total biofuels blended under the category took the form of lipid-based biodiesel (5). Biofuels under the cellulosic biofuels category are those that are derived from lignocellulosic feedstock and achieve a 60% GHG emissions reduction relative to conventional fuels. The blending mandate for this last category did not commence until 2010 under the original legislation due to the lack of any qualifying production at the time of its passage, which was not the case for the other categories (corn ethanol, cane ethanol, and soybean biodiesel, which qualify for the total

renewable fuels, advanced biofuels, and biomass-based diesel categories, respectively, have long histories of commercial-scale production) (6,7). However, the cellulosic biofuels category is to become the largest individual category by 2022 with a blending mandate of 16 billion gallons, surpassing even the cap on corn ethanol (see Figure 1.3).

The replacement of the RFS1's broad definition of a biofuel with the RFS2's strict, environment-based definitions denoted an important shift in the primary motivation underlying the mandate. Whereas the RFS1 was explicitly created as a means of improving U.S. energy security by replacing imported petroleum with domestic fuel ethanol (8), the inclusion of GHG emissions reduction thresholds in the RFS2's definitions of biofuels shifted the emphasis to environmental security. Congress placed the burden of fulfilling this motivation on the cellulosic biofuels category by defining it according to the strictest GHG emissions reduction threshold and making it the single largest category by volume of the four categories. The ultimate ability of the RFS2 to accomplish its environmental purpose will depend on the cellulosic biofuels industry's successful commercialization from no production in 2010 to 16 billion gallons of production twelve years later. By comparison, the U.S. corn ethanol industry required three decades of commercialization to achieve its first 14 billion gallons of production (1).

The RFS2 employs two mechanisms to ensure compliance with the blending mandate. The first mechanism is the imposition of a binding blending requirement on refiners and other parties supplying conventional fuels to the retail market, termed "obligated blenders" by the program. Obligated blenders are required to blend a percentage of the overall RFS2 mandate in a given year that is determined by their share of the U.S. refining market; those failing to blend the required amount face a fine of up to \$32,500 per violation

(i.e., per gallon) per day (2), ensuring that non-compliance is an economically-irrational behavior. A tradable compliance commodity known as Renewable Identification Numbers (RIN) is used to demonstrate compliance on the part of obligated blenders to the Environmental Protection Agency (EPA), which is tasked with implementing the RFS2. A RIN is created when a gallon of biofuel is produced or imported and contains information on the type of biofuel, when it was produced or imported, where it was produced or imported, etc. The RIN remains attached to the biofuel until it is blended with conventional fuel for retail, at which it is separated by the blender. Each obligated blender must submit a sufficient number of RINs to the EPA by the end of each year to demonstrate that it has fulfilled its share of the annual blending mandate.

RINs also operate as a mechanism to ensure that biofuel is either produced or imported in sufficient quantities to permit compliance with the blending mandate. As tradable compliance commodities RINs can be sold from one party to another much in the manner that carbon credits are traded in cap-and-trade schemes. For example, an obligated blender that possesses more RINs than it needs to demonstrate full compliance in a given year can sell the excess to an obligated blender holding insufficient RINs. The ability of RINs to be traded ensures that those for each category have a unique market price, creating a monetary incentive for the generation of separated RINs in addition to the aforementioned non-compliance fine via biofuel blending. Furthermore, since RINs are only created via the production or importation of biofuel, at least some of each RIN's value can be expected to be passed on to the biofuel producer, incentivizing the biofuel's production as well as its blending with conventional fuels for retail. Due to the trading mechanism RINs provide enough of a subsidy to biofuel producers and non-obligated blenders to ensure that sufficient

biofuel is produced and blended with conventional fuels to make compliance with the RFS2 possible but without creating windfall profits (9). RINs increase in value when the costs of biofuel production and/or blending rise and vice versa, falling to zero when blending exceeds the volumetric mandate since no RIN value is necessary at that point to ensure compliance with the mandate. Because RIN values are determined via a market-based mechanism, however, they can only be known for those biofuel categories in which biofuel is being both produced and blended with conventional fuels for retail.

Finally, the value of a RIN depends on the type of biofuel produced (as distinct from the RFS2 category that it qualifies for). RINs are attributed on an ethanol-equivalent basis for the four RFS2 categories. The different types of biofuels that have been blended with conventional fuel as part of the RFS2 mandate include ethanol, FAME biodiesel, gasoline, and diesel fuel (5). The energy content of each biofuel type is different (10) and FAME biodiesel, gasoline, and diesel fuel receive more than one RIN per gallon as a result (see Table 1.2), making them more valuable than ethanol in terms of RIN value per gallon. The relationship between RIN value and biofuel type is limited to the aforementioned RFS2 categories and the biofuel's energy content, however; so long as a biofuel achieves the 60% GHG emissions reduction threshold of the cellulosic biofuel category then there is no additional value to be attained by achieving a still-higher reduction threshold (assuming that cellulosic biofuel RINs have a greater value than those of the other categories).

Type of biofuel	Number of RINs created per gallon	
Ethanol	1.0	
Biogas	1.0	
Heating oil	1.1 – 1.6	
Butanol	1.3	

Table 1.2. RINs created per type of biofuel (5)

Table 1.2 continued			
Type of biofuel	Number of RINs created per gallon		
FAME biodiesel	1.5		
Gasoline (naptha)	1.5		
Diesel fuel	1.5 - 1.7		

The total renewable fuels, "other" advanced biofuels (the share not attributable to the biomass-based diesel and cellulosic biofuels categories), and biomass-based diesel volumetric mandates have largely been met to date (11,12) and even exceeded in the case of the latter (5). Production of cellulosic biofuels, on the other hand, has fallen far short of the category's volumetric mandates since the RFS2's inception due to a lack of capacity investment. Recognizing that insufficient global capacity has been available to meet the mandate, the EPA has adjusted the cellulosic biofuels volumetric mandates down by as much as 99% in every year since 2010 (see Figure 1.4). Actual production of cellulosic biofuels still fell short of the EPA's adjusted volumes by at least 94% and as much as 100% for each of those years.



Figure 1.4. EPA adjustments to the cellulosic biofuels volumetric mandates, 2010-2014 (2). Actual volume for 2014 is not known at time of writing. Actual volumes for 2012 and 2013 were 0.02 million and 0.82 million gallons, respectively.

The lack of cellulosic biofuel production to date has created an implementation dilemma for the EPA and raised uncertainty regarding the category's future viability. In calculating the adjusted cellulosic biofuels volumetric mandates the EPA has intentionally erred towards overestimates in an effort to support investment in the industry (13,14). This practice has been employed since a biofuel category's RIN values approach zero when blending exceeds the volumetric mandate. Were the EPA to release an adjusted volumetric mandate that is ultimately surpassed by blending then cellulosic biofuel RINs would fall in value despite the original volumetric mandate not being met, inhibiting continued capacity investment. By setting the adjusted volumes too high, however, the EPA has placed obligated blenders in the position of being legally required to blend non-existent biofuels with conventional fuels for retail. In January 2013 the U.S. Court of Appeals for the District of Columbia overturned the cellulosic biofuels mandate due to the EPA's practice of overestimating the adjusted volumes, although it limited the scope of its ruling to the volumetric mandate in 2013 (14).

Notwithstanding the EPA's efforts, the complete lack of cellulosic biofuel production in 2010 and 2011 and virtual lack of production in 2012 have prevented the market from identifying a cellulosic biofuels RIN price due to a lack of RIN supply. This has prevented cellulosic biofuels RINs from successfully fulfilling one of their primary purposes, which is to incentivize capacity investment by providing producers and blenders with a subsidy for ensuring that the biofuel is made available for blending and retail. Investment has been inhibited by the lack of a clear subsidy for cellulosic biofuel production, in turn ensuring that the subsidy continues to be missing in the future due to a lack of investment and subsequent production. Biofuels qualifying for the other three RFS2 categories were already being produced and/or imported at the time that the respective volumetric mandates were implemented, allowing for the value of each category's RINs to be quickly known. This discrepancy between the categories will only grow more pronounced as the cellulosic biofuels volumetric mandate continues to increase rapidly on an annual basis, with the 3 billion gallons mark to be reached in 2015.

The future viability of the broader RFS2 also became more uncertain in 2013 due to technical issues involving the blending and consumption (as opposed to production) of fuel ethanol. Unlike hydrocarbon-based fuels, ethanol is miscible with water and any water contaminating a pipeline, storage tank, or pump (e.g., via condensation) is absorbed by ethanol as it passes through (10). In addition to causing corrosion in unmodified equipment, blending ethanol that has absorbed water with gasoline results in phase-separation that interferes with proper engine operations. U.S. ethanol blends with gasoline in unmodified

vehicles have largely been limited to 10 vol% ("E10") because of this miscibility; while the EPA began permitting blends of up to 15 vol% ("E15") in 2010 (15), its acceptance by consumers has been minimal and limited to the Midwest (16). Adoption rates of flex-fuel vehicles, which can consume ethanol blends of up to 85 vol% ("E85"), have also been too low in the U.S. to cause a significant increase in ethanol consumption (17). Ethanol consumption in the U.S. is roughly limited to 10 vol% of gasoline consumption with the current fuel infrastructure as a result.

The RFS2 was created at a time when U.S. gasoline consumption was forecast to steadily increase over the next two decades (18). Long-term gasoline consumption is now forecast to slowly decline over the same period due to changing driving habits, the slow economic recovery in the aftermath of the 2008 Financial Crisis, and significant improvements in vehicle fuel economy (19). In 2013 the U.S. hit the ethanol "blend wall", which is defined as the point at which ethanol consumption equals 10 vol% of gasoline consumption, when ethanol production continued to increase even as gasoline consumption fell. The arrival of the blend wall has created multiple problems for the RFS2. The first is over the ability of U.S. refiners to blend and consumers to utilize the volumes of ethanol required by the mandate. In 2017 only 13 billion gallons of ethanol will be permitted by the blend wall according to current forecasts (19), which is below the 15 billion gallons of corn ethanol permitted by the mandate, let alone any additional volumes of ethanol in the form of advanced and cellulosic biofuels. The type of biofuel produced in the future within the cellulosic biofuels category is now an important question with regard to the category's future feasibility. If the biofuels produced under the category primarily take the form of ethanol, as was originally anticipated by Congress, then corn, cane, and cellulosic ethanol will be forced

to compete for a share of a shrinking ethanol market (see Figure 1.5). Hydrocarbon-based biofuels (and butanol to a lesser extent) are not limited by the ethanol blend wall due to their chemical similarity to petroleum-derived fuels, however, so cellulosic biofuels taking those forms will not face the same future constraints as cellulosic ethanol.



Figure 1.5. The E10 blend wall relative to the RFS2 volumes that do not exclude fuel ethanol (2,19).

The arrival of the blend wall in 2013 also increased uncertainty regarding the future viability of the RFS2 by causing RIN prices, particularly for the total renewable fuel category, to increase by 2800% between January and July of that year. RIN prices for the total renewable fuels (D6) category had barely exceeded transaction costs from 2010 to the end of 2012 due to the ease with which the corn ethanol industry produced sufficient biofuel for its share of the blending mandate to be met (20). Never exceeding \$0.04 during that period, D6 RIN prices were too low to incentivize blending by many obligated blenders as they found it easier to purchase the RINs necessary to demonstrate their compliance on the

market. The lack of volatility in the RIN markets changed in early 2013 following the broad realization that the volumetric mandates for the year exceeded the ethanol blending volumes permitted by a 10 vol% blend wall. Faced with insufficient blending capacity and consumers unwilling to purchase higher ethanol blends, obligated blenders were required to continue purchasing RINs even as their prices rapidly increased (see Figure 1.6). Large financial institutions reportedly increased the volatility by purchasing large volumes of RINs at low prices in anticipation of the blend wall and then waiting to sell them to obligated blenders until after they had greatly increased in value (21). One large obligated blender, Valero, reported over the summer that it expected its total RIN costs for the year to increase to \$800 million due to the rising prices.



Figure 1.6. RIN prices since May 2013 (22). D4 = Biomass-based diesel category; D5 = Advanced biofuel category; D6 = Total renewable fuel category.

The refining industry launched a large lobbying effort in Washington D.C. during the second half of 2013 to convince Congress and the EPA to reform or even repeal the RFS2 as

a means of reducing RIN prices. In August the EPA responded to this effort by promising to consider the constraints imposed by the ethanol blend wall in determining the volumetric mandates for 2014 (23), causing RIN prices to begin falling in value (see Figure 1.6). In November the EPA formally proposed to adjust the volumetric mandates for the total renewable fuels and advanced biofuels categories, much as it has done to the cellulosic biofuels category, so that the combined volume of both does not exceed 10 vol% of U.S. gasoline consumption (Table 1.3). While the EPA has yet to release its final rulemaking on the 2014 volumetric mandates, such a reduction would establish the precedent of reducing the total blending requirement for the two categories each year for the foreseeable future (see Figure 1.5). Any cellulosic biofuels production taking the form of ethanol would presumably be constrained in a similar manner, forcing it to compete with corn ethanol and cane ethanol for both RINs (since the EPA would determine the annual allocation) and market share. Cellulosic hydrocarbon-based biofuels face no such limitation, however, and future production of this biofuel type would not be limited even in the event that the EPA formalizes the blend wall constraints within the RFS2's volumetric mandates.

Category	Original 2014 volumes (billion gallons)	Proposed 2014 volumes (billion gallons)
Total renewable fuels	18.15	15.21
Corn ethanol cap	14.40	13.01
Advanced biofuels	3.75	2.20
Biomass-based diesel ^a	1.28	1.28
Cellulosic biofuels	1.75	0.02
"Other" advanced biofuels	0.72	0.26

Table 1.3. 2014 volumetric mandates for the RFS2, both original and proposed (24).

^aAll volumes except biomass-based diesel are presented on an ethanol-equivalent basis; each gallon of biomass-based diesel is equivalent to 1.5 gallons of ethanol.

Forecasting the future viability and results of the RFS2 under the ethanol blend wall requires knowledge on the feedstocks and pathways likely to be employed and resulting biofuel types. Cellulosic ethanol's production costs are significantly higher than those of both corn ethanol and cane ethanol due to high feedstock costs and the recalcitrant nature of lignocellulose (6). Cellulosic ethanol is uncompetitive with other ethanol pathways (let alone petroleum) on an unsubsidized basis as a result. The future viability of the cellulosic biofuels mandate under a scenario in which cellulosic ethanol is the category's primary fuel type would require RINs to subsidize its production, blending, and consumption at a sufficient level to make it competitive with these other products (25). The lack of a blend wall for cellulosic hydrocarbon-based biofuels reduces the level of subsidization necessary under a scenario in which they are the cellulosic biofuels category's primary fuel type since only production would need to be incentivized, although this would also be determined by the production costs of the respective pathways.

Knowledge of these three factors (feedstocks, pathways, and biofuel type) is also important if the impacts, both positive and negative, of the cellulosic biofuels mandate are to be determined. Cellulosic biofuels differ from corn and cane ethanol in that the latter are almost entirely derived from single feedstocks in specific regions (e.g., the U.S. Midwest, Brazil's Mid-South and Northeast). It has been estimated that cellulosic biofuels production in the U.S. alone will utilize three broad feedstock groups (forest biomass, agricultural residue, and dedicated energy crops) from at least four regions (26). Furthermore, in 2012 no fewer than six different cellulosic biofuel pathways within both the biochemical and thermochemical platforms producing both ethanol and hydrocarbon-based biofuels were expected to achieve commercial-scale production by the beginning of 2015 (27). Techno-

economic analyses (TEA) of these pathways have found that their production costs vary widely even after accounting for differences in assumed feedstock costs and product market prices (28–31). Furthermore, production costs for an individual pathway are determined in part by the location in which it is employed due to regional differences in feedstock type, feedstock cost, capital costs, and market prices (32).

Differences in the ability of each cellulosic biofuel pathway to compete with conventional fuels will also contribute to other impacts of the RFS2's cellulosic biofuels mandate that are not explicitly economic in nature. These impacts can be categorized as environmental, ecological, and socioeconomic. The environmental impact that has received the greatest attention due to the way in which the RFS2 is structured is an individual pathway's reduction to lifecycle GHG emissions relative to conventional fuels. While several of the pathways currently undergoing commercialization have been determined to achieve a 60% GHG emissions reduction threshold (33–38), the actual emissions reduction varies by pathway and feedstock. Hsu et al. (39) identifies differences in feedstock yields, nitrogen fertilizer application rates, removal rates, and moisture contents between different lignocellulosic feedstocks as being the most influential parameters in terms of lifecycle GHG emissions for a single cellulosic biofuels pathway. Hsu (35) calculates a 65% reduction to GHG emissions for the fast pyrolysis and hydroprocessing pathway and a 90% reduction for the gasification and mixed alcohols synthesis pathway, both of which yield cellulosic biofuels. Zhang et al. (34) reports that the 65% reduction for the fast pyrolysis and hydroprocessing pathway can be increased to an 88% reduction by utilizing biomass-based hydrogen rather than natural gas-based, although Wright et al. (28) calculates that this increases the pathway's production costs by as much as 50%. Tao et al. (40) calculates that

biofuel type also affects the lifecycle emissions reduction of a cellulosic biofuels pathway, with ethanol produced via dilute-acid pretreatment and enzymatic hydrolysis generating a 69% reduction versus a 63% reduction for isobutanol produced via the same route.

The EPA (41,42) calculates an emissions reduction range of 65-129% for the cellulosic biofuel pathways that it has approved under the RFS2 to date, indicating that some pathways are carbon negative (i.e., they achieve a net reduction to atmospheric GHGs) when the appropriate product portfolio is generated. In some cases a trade-off is identified between GHG emissions reductions and biofuel yields (34,43). The type of land on which the lignocellulosic feedstock is produced is also an important factor: Searchinger et al. (4) reports a 50% increase to lifecycle GHG emissions for cellulosic ethanol relative to conventional fuels when switchgrass feedstock is grown on productive cropland in the U.S. Midwest, while Gelfand et al. (44) reports large emissions reductions when the feedstock is instead grown on marginal cropland in the same region.

The ecological impacts of cellulosic biofuels are primarily associated with direct land-use change. Positive ecological impacts can result from the growth of certain types of lignocellulosic biomass on marginal lands; for example returning marginal cropland in the Midwest to its original state via reforestation or conversion to switchgrass production. Baker et al. (45) and Elobeid et al. (46) both calculate that the reforestation or afforestation of productive cropland would cause U.S. food production to fall and increase the prices of crops such as corn, soybeans, and wheat by as much as 39%, however, making the type of land on which lignocellulosic feedstock production occurs an important factor. Wildlife in production areas can also be impacted by these decisions, as some species prefer habitats consisting of long post-harvest stubble while others prefer short stubble (47). In the case of corn stover,

high harvest rates contribute to soil erosion from water and wind as well as soil carbon depletion (48). Corn stover yields are closely correlated to grain yields (49) and increased future demand for stover feedstock could result in higher rates of fertilizer application in an effort to boost both. Finally, future demand for stover could also induce farmers to switch from corn-soybean rotations to continuous corn crops (50), much as occurred in the U.S. during the first decade of the 21st century, potentially causing greater losses of nitrogen and phosphorous to water (51).

Finally, cellulosic biofuel production can be expected to have socioeconomic impacts within the U.S. Whereas the costs of the RFS2 are borne by conventional fuel consumers via the refiners that are required to either purchase or generate separated RINs, the immediate beneficiaries of the mandate have yet to be identified. Many U.S. lignocellulosic feedstocks are only grown in certain regions (26), making the choice of both feedstock and pathway important determinants of the regions in which feedstock and biofuel production under the RFS2 will ultimately occur. Studies report mixed socioeconomic impacts from existing bioenergy projects including regional revitalization (52), concentration of wealth in large-scale producers resulting in the social exclusion of small producers (53), and shifting attitudes toward bioenergy adoption (54). A better understanding of the type and magnitude that these impacts are likely to have under the RFS2 requires further knowledge on the cellulosic biofuel feedstocks and pathways that are most likely to contribute to the mandate.

1.2. Dissertation Objectives

The primary objective of this dissertation is to develop a methodology for comparative techno-economic and uncertainty analyses of cellulosic biofuel pathways. This primary objective is completed via three specific objectives:

- Quantify the economic competitiveness via techno-economic analysis of eight cellulosic biofuel pathway scenarios within two energy commodity price scenarios under price volatility.
- Develop a framework for quantifying the economic competitiveness of cellulosic biofuel pathways by incorporating pathway-specific financial assumptions and energy commodity price uncertainty into techno-economic analysis.
- Quantify the economic competitiveness via techno-economic analysis of eight cellulosic biofuel pathway scenarios with pathway-specific financial assumptions, energy commodity price uncertainty, and region-specific location scenarios.

1.3. Intellectual Merit

This dissertation advances the current methodology employed to quantify the economic competitiveness of cellulosic biofuel pathways by incorporating realistic financial assumptions, price volatility and uncertainty based on the combination of historical price movements and price forecasts, and region-specific factors into a single framework. The results of the studies included in this dissertation provide new information on the short-term competitiveness of the cellulosic biofuel pathways currently being commercialized relative to petroleum and one another. Furthermore, the methodology employed to produce this result has been presented in a manner that can be utilized by researchers with access to Monte

Carlo simulation software, public databases, and the open literature. This methodology can incorporate new data on cellulosic biofuel pathways and commercialization as it becomes available to continuously update the competitiveness assessment and provide additional information on the likely near- and mid-term composition of the cellulosic biofuel industry, permitting research into the future viability and impacts of the industry broadly and the RFS2 specifically.

1.4. Dissertation Organization

This dissertation is organized as five chapters. Chapter 1 provides a brief history of U.S. biofuels mandates in the 21st century and an overview of the specific mechanisms by which the RFS2 incentivizes sufficient biofuel production to ensure compliance with its blending mandates. Chapter 1 also discusses both the challenges facing the future viability of U.S. cellulosic biofuels and the quantification of the same. Finally, Chapter 1 details the objectives, intellectual merit, and organization of this dissertation.

Chapter 2 addresses this dissertation's first objective, which is to quantify the economic competitiveness of eight cellulosic biofuels pathways within two 20-year energy commodity price scenarios under price volatility. Chapter 2 develops each cellulosic biofuel pathway based on the results of techno-economic analyses in the open literature. The energy commodity price scenarios are derived from government price forecasts from before and after the large-scale U.S. exploitation of shale gas reserves. Price volatility is modeled via a Monte Carlo simulation methodology that derives probability distributions from historical monthly variations around the annual mean prices of five energy commodities. These probability distributions are applied to both energy commodity price scenarios, allowing for

the economic competitiveness of each pathway scenario to be calculated under future price volatility. Chapter 2 has been drafted as a standalone manuscript that has been published in the refereed international journal *Fuel* and includes supporting information.

Chapter 3 addresses the second objective of this dissertation, which is to develop a framework for quantifying the economic competitiveness of cellulosic biofuel pathways. Chapter 3 accomplishes this by incorporating pathway-specific financial assumptions and energy commodity price uncertainty into techno-economic analysis. The pathway-specific financial assumptions are calculated using methodologies from the financial analysis literature, replacing the generic assumptions currently employed by techno-economic analyses of bioenergy pathways. Energy commodity price uncertainty employed by techno-economic simulation on the basis of each commodity's historical month-on-month price variations and 20-year price forecasts. Chapter 3 has been drafted as a standalone manuscript that has been submitted for publication in the refereed international journal *Biofuels*.

Chapter 4 addresses the third objective of this dissertation, which is to quantify the economic competitiveness of eight cellulosic biofuel pathway scenarios with pathway-specific financial assumptions, energy commodity price uncertainty, and region-specific location scenarios. Chapter 4 expands upon the methodologies introduced in Chapters 2 and 3 by incorporating region-specific factors such as differences in capital costs, feedstock type, feedstock costs, commodity prices, and tax rates. Eight location scenarios representing planned commercial-scale cellulosic biofuel projects in the U.S. are developed and their economic competitiveness with conventional fuels and one another are quantified and compared. Chapter 4 has been drafted as a standalone manuscript for submission to a refereed international journal.

Chapter 5 summarizes the primary conclusions of this dissertation as presented in Chapters 2-4. It also discusses future research directions that would expand upon and utilize the methodologies introduced in the previous chapters.

CHAPTER 2. TECHNO-ECONOMIC IMPACTS OF SHALE GAS ON CELLULOSIC BIOFUEL PATHWAYS

A paper published in the international refereed journal Fuel

Tristan R. Brown¹ and Mark M. Wright²

2.1. Abstract

This analysis quantifies the economic feasibility of cellulosic biofuel pathways under fossil fuel price uncertainty. Eight pathway scenarios are developed on the basis of existing techno-economic analyses and projected fossil fuel commodity prices from the Energy Information Administration's (EIA) 2010 Annual Energy Outlook (AEO). A 20-year net present value (NPV) is then calculated for each pathway scenario. Uncertainty distributions are developed for each pathway scenario by fitting historical monthly price variance distribution curves for each fossil fuel commodity to their projected annual prices. Finally, a sensitivity analysis is completed by replacing the EIA's AEO 2010 projected prices with those from its AEO 2013, the latter incorporating recent exploitation of U.S. shale gas reserves into its projections. The results of this analysis indicate that fast pyrolysis scenarios see the greatest increase in estimated NPV value followed by gasification and acetic acid synthesis scenarios. Fischer-Tropsch synthesis scenarios remain largely unaffected by the updated EIA projections. Methanol-to-gasoline and enzymatic hydrolysis NPVs decrease as a result of lower projections for fossil fuel prices.

Keywords³

¹ Graduate student in Bioeconomy Institute, Iowa State University; primary researcher and author.

² Assistant professor in Department of Mechanical Engineering, Iowa State University; corresponding author.

 $^{^{3}}$ AAS = acetic acid synthesis; AEO = Annual Energy Outlook; DCFROR = discounted cash flow rate of return; DHG = direct-heat gasification; EH = enzymatic hydrolysis; EIA = Energy Information Administration; FPH = fast pyrolysis and hydroprocessing; FTS = Fischer-Tropsch synthesis; IHG = indirect-heat gasification; HTG = high-temperature gasification; IRR = internal rate of return; LCA = life cycle assessment; LPG = liquefied

Cellulosic biofuel; shale gas; techno-economic analysis; uncertainty analysis

2.2. Introduction

The widespread deployment of technology enabling the inexpensive extraction of shale gas has caused U.S. natural gas production to increase substantially in recent years, with annual U.S. production currently at a level never seen before (55). One effect has been significantly lower U.S. monthly wellhead gas prices, which in April 2012 fell below \$2/MMBTU for the first time in the 21st century (56). This development signifies a fundamental shift in the dynamic between the prices of petroleum and natural gas.

Natural gas has historically been a byproduct of petroleum extraction, resulting in a strong correlation between the prices of the two commodities (56,57). Divergences in their price movements over the last 25 years have been characterized by their brevity and infrequency, usually corresponding to extreme weather events and rarely lasting for more than a few months (see Figure 2.1). Starting in 2009, however, this correlation in price movements began a monotonic divergence. By 2012 the price of WTI petroleum had increased 200% over the previous decade on a nominal basis, while the wellhead price of U.S. natural gas on the same basis had fallen to a 13-year low. Meanwhile both U.S. natural gas production and proved reserves have reached historical highs (58), with the increase in the latter almost entirely attributed to an increase in proved reserves of shale gas (59).

petroleum gas; LTG = low-temperature gasification; MMBTU = million British thermal units; MT = metric ton; MTG = methanol to gasoline; NCG = non-condensable gases; NPV = net present value; RFS2 = revised Renewable Fuel Standard; RIN = Renewable Identification Number; SMR = steam methane reforming; TEA = techno-economic analysis; TPEC = total purchased equipment cost; TPI = total project investment; WTI = West Texas Intermediate.



Figure 2.1. Monthly U.S. natural gas wellhead and WTI contract nominal price, 1983-2012 (56,57).

The EIA forecasts this price divergence to be the start of a new multi-decade trend (see Figure 2.2). The U.S. natural gas wellhead price in 2035 is currently projected to be only 58% that of the commodity's trend based on the historical correlation between U.S. natural gas and WTI prices (19). Should these projected prices occur then this new price relationship will represent a complete reversal of the historical relationship.


Figure 2.2. Historical (1997-2012) and projected (2013-2040) increases in annual nominal natural gas (\$/MMBTU) and West Texas Intermediate (WTI) petroleum (\$/bbl) spot prices (19).

The wide gap between projected natural gas and petroleum prices has caused some U.S. policymakers to question the wisdom of investing in high-cost cellulosic biofuels at a time when domestic natural gas is available as an inexpensive feedstock for production of fossil fuel-based synthetic transportation fuels (synfuels). Congressional legislation introduced in 2012 would expand the RFS2 to include both biofuels and synfuels (60). Two would-be cellulosic biofuel producers announced in the same year that they were switching from biomass to natural gas as alternative fuels feedstock in part due to low natural gas prices (61,62). The domestic economics of synfuel pathways are more attractive at present than in the past due to lower input costs (natural gas and coal) and higher output prices (gasoline and diesel fuel). While lifecycle assessments (LCA) of synfuels pathways such as gas-to-liquids (GTL) and coal-to-liquids (CTL) report higher greenhouse gas (GHG) emissions than for petroleum-based transportation fuel pathways (63–65), the absence of a national carbon tax or price program in the U.S. limits the negative impact that this has on the fuels' economic feasibility.

Largely ignored in the discussion of the impact of shale gas on the economics of alternative transportation fuel production is its potential ability to improve the economic feasibility of biofuel production. U.S. biofuels policy has undergone a major shift over the last decade, greatly expanding its scope to include hydrocarbon-based cellulosic biofuels in addition to cellulosic ethanol (66). Hydrocarbon-based biofuel pathways directly utilize hydrogen in either one- or two-step processes of deoxygenation and depolymerization to increase overall yields of monomeric hydrocarbons. Hydrogen can be derived from a number of sources although the least expensive source with current technology is produced via the SMR of natural gas (67,68). The new relationship between natural gas and petroleum prices can therefore be expected to directly impact the economic feasibility of hydrocarbon-based biofuel pathways.

Based on current facility construction, U.S. cellulosic biofuel production will reach 215 MM gallons gasoline-equivalent in 2014, slightly more than half of which will be hydrocarbon-based (27). These cellulosic biofuel facilities will employ several different pathways, including gasification and FTS; gasification and MTG; and EH and fermentation. This analysis quantifies the impact of U.S. shale gas production on the economic feasibility of these pathways, in addition to fast pyrolysis and hydroprocessing and gasification and AAS. Spreadsheet models were created using pathway data in the techno-economic literature to quantify the economic feasibility of each pathway under two different economic scenarios based on the EIA's price forecasts from its AEO 2010 and AEO 2013. Uncertainty analysis

was employed using historical price variation data in conjunction with the price projections to quantify scenario performance under uncertainty.

2.3. Methodology

This study proceeds as follows: (1) process and economic data of select biofuel pathways were collected and adjusted, (2) historical data on monthly commodity price variation distributions were gathered and fit to probability distributions, (3) a range of NPVs were estimated for each pathway scenario based on stochastic analysis of the commodity prices and economic parameters. Figure S2.1 outlines the steps taken in this study.

Seven different cellulosic biofuel pathways were selected to develop eight pathway scenarios: (1) HTG and FTS; (2) LTG and FTS; (3, 4) stand-alone FPH; (5) DHG and AAS; (6) IHG and AAS; (7) EH and fermentation; and (8) gasification and MTG synthesis. The two gasification and FTS scenarios were developed using data from the high- and low-temperature scenarios presented by Swanson et al. (2010) (29). Two separate TEAs of the FPH pathway were used to develop Scenarios 3 and 4: Brown et al. (2013) (69) and Jones et al. (2009) (70) (referred to here as "BFPH" and "JFPH", respectively). The direct- and indirect-heat gasification scenarios presented by Zhu and Jones (2009) (71) were used to develop the two gasification and AAS scenarios. The EH and fermentation scenario was based on work by Kazi et al. (2010) (72). Finally, the gasification and MTG scenario was developed using the results of Phillips et al. (2011) (73,74). The pathway scenarios are described in more detail in a later section. Table S2.1 summarizes the mass balances and annual operating hour assumptions of the eight pathway scenarios. All of the pathway scenarios are based on original facility capacities of 2000 MT/day, although it should be

noted that optimal facility sizes can differ across pathways (75). The pathway scenarios were chosen based on the availability of detailed techno-economic analysis (TEA) results and this selection is not an exhaustive list. However, the methods in this study can be readily applied to TEAs of emerging pathways as they are published provided they contain the TEA data presented in Table S2.1 and Table S2.2.

While the TEAs used to develop each scenario in this analysis are similar in that the results are presented in formats that are broadly comparable, there are several differences between them (e.g., cost basis year, capital cost calculation methodology, annual operating hours, etc.). First, the assumed feedstock cost is not the same across all pathway scenarios and has been adjusted to \$83/MT in 2011 dollars in this analysis. Second, while differences in the cost basis year are adjusted to 2011 dollars using the U.S. city average Consumer Price Index as the cost multiplier (76) (see Table S2.2), the capital cost calculation methodology and annual operating hours remain the same as presented in the cited studies. While these assumptions affect the results of each scenario, it is not immediately clear whether or not they are arbitrary in nature. For example, while the companion technical report to Phillips et al. (74) provides a detailed justification of its capital cost calculation methodology (73), the remaining studies do not. Given the ambiguous nature of the capital cost calculations, then, this analysis does not adjust the capital cost calculation methodologies and annual operating hours of the cited studies to place them all on the same basis. Finally, the capacity factors for each pathway scenario also remain unchanged from the cited studies for the same reason, although this too can be expected to affect pathway economic feasibility (77,78). The results of each pathway TEA are not strictly comparable as a result.

Several different energy commodities are employed as either inputs or outputs by the eight pathway scenarios (see Table 2.1 and Table S2.1). While the TEAs listed in Table S2.1 and Table S2.2 assume static prices for each commodity over the 20 year period analyzed, these prices have historically been subject to substantial price volatility (see Figure S2.2). Assuming that the commodity prices remain volatile in the future, this volatility makes it unlikely that static prices will occur in the future. It is more likely that the future prices will exhibit similar volatility. Rather than calculate a point estimate of pathway economic feasibility as was done in the previously-cited TEAs, then, this analysis employs Monte Carlo simulation to calculate a range of NPVs and probabilities based on price variability for the commodities employed as inputs and outputs.

	r	price (79).		1
Commodity	Distribution	2010 Projection	2013 Projection	Historical
		(2013-2032	(2013-2032	data years
		average)	average)	and source
Electricity	Triangular	\$0.066/kWh	\$0.065/kWh	2001-2012
(industrial)				(29)
Diesel fuel	Logistic	\$3.18/gal	\$3.30/gal	1994-2012
	_		_	(30)
Gasoline	Logistic	\$3.09/gal	\$2.95/gal	1993-2012
				(30)
LPG	Logistic	\$2.02/gal	\$1.26/gal	1993-2012
				(31)
Natural gas	Logistic	\$7.73/MMBTU	\$4.49/MMBTU	1992-2012
(Henry Hub)				(2)
Hydrogen ^a	Logistic	\$1.76/kg	\$1.08/kg	-

Table 2.1. Probability distribution function selection and expected value for each commodity price (79).

^aCalculated as a function of natural gas price (80)

Electricity, diesel fuel, gasoline, hydrogen (produced from natural gas via SMR), LPG, and natural gas were selected for the Monte Carlo simulations due to their use in biorefineries, commercial availability, and the availability of historical monthly price data. Feedstock, catalysts and process water were excluded despite their use as inputs in most of the pathway scenarios due to a lack of available historical data for the first two and a low impact on pathway economic feasibility for the latter. Feedstock type, ultimate analysis, and moisture content are presented in Table S2.3. Other important sources of pathway cost uncertainty are beyond the scope of this study and discussed elsewhere (81,82).

A variety of methodologies for developing energy commodity price distributions have been employed in the techno-economic literature. The simplest technique represents their uncertainty via triangular distributions based on either a historical price range (83) or a predetermined percentage range (e.g, +/- 30%) around a mean projected price (84–86). While straightforward, Shlyakhter et al. (1994) report that these price distributions are inaccurate when applied to energy markets since they attribute zero probability to outliers (87).

A second technique represents energy commodity price uncertainty via uniform distributions based on the combination of historical price ranges and contemporary market prices, with the former operating around the latter (88,89). This technique attributes equal probabilities to all prices within the range and makes the important assumption that historical price ranges are an accurate predictor of future price ranges. While true for some commodity prices in the past (e.g., WTI petroleum from 1983 to 2003), this assumption is unsuitable for an analysis considering the breakdown of an historical price relationship. Furthermore, this technique also assumes that the future mean price is the same as the contemporary price, which is unsuitable when future commodity market conditions are anticipated to be substantially different than contemporary conditions.

A third technique represents energy commodity price uncertainty via normal distributions based on projected price ranges and projected mean prices (90). While this

allows for anticipated changes in future commodity market conditions to be accounted for, normal distributions around projected prices have also been found to underestimate the frequency of extreme outcomes (although not to the same extent as triangular distributions) (87).

This analysis accounts for price outliers by developing and employing fitted distributions for each input parameter (see Table 2.1). The distributions were developed using the historical monthly price variations of each energy commodity in a given year around the annual average price for the same year. The appropriate distribution for each commodity price was selected based on the results of the Anderson-Darling goodness-of-fit test (91). The monthly price variation distributions are then applied on a monthly basis to the respective annual commodity prices in the DCFROR spreadsheet. This approach generates logistic distributions for all of the historical commodity price variations except electricity, reflecting heavier tails relative to normal distributions. The historical data generates a triangular distribution as the best fit for electricity. An important assumption of this technique is that the historical price variation of each commodity remains the same in the future.

Reference operating costs were also adjusted for each scenario to account for differences in assumed commodity costs. Prices for all major commodities (the averages are presented in Table 2.1) were derived from the annual prices projected for 2013-2032 by the EIA in its AEO 2010 report (79). The prices of gasoline and diesel fuel exclude fuel excise taxes by subtracting the national average excise tax for both (92) from the EIA's projected prices. It is assumed that hydrogen is produced on site via SMR of natural gas. Hydrogen price was calculated using the Department of Energy's (DOE) H2A Central Hydrogen

Production Model (80) as a function of the natural gas price and a 149 MT/day output capacity.

A stochastic DCFROR spreadsheet was used to calculate a 20-year NPV probability distribution for each pathway scenario using the data presented in Table S2.1 and Table S2.2. The stochastic model is based on a deterministic DCFROR model developed by the National Renewable Energy Laboratory and modified by Wright et al. (2010) (93,94). The stochastic model has two important differences from the original deterministic model. First, the stochastic model operates on a monthly rather than annual basis to permit the incorporation of the monthly price variation distributions. Second, the stochastic model incorporates Monte Carlo simulations to develop projected commodity price distributions for each of the commodities presented in Table 2.1 and the 240 months accounted for by the spreadsheet. All NPV calculations assume a 10% IRR, 100% equity financing, and a 35% corporate income tax rate. Of these three assumptions, the NPV calculations are most sensitive to the assumed IRR (77,78). The Monte Carlo simulations were employed to obtain NPV distributions and quantify the certainty that the NPV of each pathway scenario is > \$0, based on the above assumptions. The simulations were performed with Crystal Ball[®] software and were based on 10,000 trials.

The historical price data was analyzed to identify correlations between different commodities. For example, gasoline and diesel fuel are both produced via the refining of petroleum and the prices of both can be expected to be closely correlated as a result. Crystal Ball® software was also used to identify correlations between historical monthly commodity prices via Spearman's correlation coefficient. Very strong correlations, defined here as a correlation coefficient of greater than 0.9, were identified for gasoline and diesel fuel prices

(0.975) and gasoline and LNG prices (0.935). These correlations were incorporated into the appropriate assumptions within the Monte Carlo simulations to prevent ahistorical price divergences between closely-correlated commodities.

Finally, a sensitivity analysis was employed by comparing the eight pathway scenarios under the AEO 2010 energy commodity price projections with those from the AEO 2013. The AEO 2013 scenario employed the same methodology as the AEO 2010 scenario, with price variability distributions derived from historical data applied to the projected prices. The AEO 2013 prices are different from the AEO 2010 prices, however, particularly for natural gas and LPG (see Figure S2.3). The sensitivity analysis quantifies the impact that the changing price relationship resulting from increased shale gas production has on the economic feasibility of the eight pathway scenarios analyzed here. A Supporting Information document containing descriptions and schematics (see Figure S2.4) of each of the pathway scenarios considered by this analysis can be found online.

2.4. Results and Discussion

4.1.1. Uncertainty Analysis Based on 2010 Projections

We compare the effects of electricity, gasoline, diesel fuel, LPG, and natural gas prices on the NPV of eight techno-economic scenarios based on seven cellulosic biofuel pathways. Figure 2.3 presents the NPV probability distributions of the economic scenarios derived from the EIA's AEO 2010/2013. The BFPH, JFPH, and gasification and MTG scenarios are all calculated as having 100% certainty of achieving NPVs of greater than zero (see Figure 2.3). On the other hand, the remaining scenarios all have a zero probability of achieving a positive NPV based on the model assumptions and projected prices.



Figure 2.3. NPV comparison of biorefinery scenario uncertainty analysis based on 2010/2013 fuel price projections. Light shading indicates the AEO 2010 scenario and dark shading indicates the AEO 2013 scenario.

The wide range in NPV standard deviations for the various gasification pathways indicates that the fuel synthesis route employed has a noticeable effect on pathway economic feasibility, as these scenarios all convert the biomass feedstock to syngas via a gasification step. The standard deviation of each pathway scenario NPV is also driven by the number of different pathway scenario inputs and their quantities. The IHG & AAS and DHG & AAS scenarios have the largest NPV standard deviations of \$54.7 MM and \$68.2 MM, respectively. The two pathway scenarios are also the largest net consumers of electricity, hydrogen, and/or natural gas (see Table S2.1). Exposure to multiple fossil fuel inputs results in a broader NPV probability distribution than those pathway scenarios with less exposure, especially when the inputs are not correlated. While this diversity of inputs decreases the sensitivity of NPV to a sharp change in the price of a single input, it also increases NPV uncertainty. However, we note that there are other factors unaccounted for by this analysis that could influence the profitability of these pathway scenarios.

4.1.2. Uncertainty Analysis Based on 2013 Projections

A sensitivity analysis is employed to quantify the effect of changing energy commodity prices on each pathway's NPV under uncertainty. This sensitivity analysis is accomplished by replacing the 20-year energy commodity prices from the EIA's AEO 2010 with those from the AEO 2013 (see Table 2.1) (19). The 20-year average electricity and gasoline prices are slightly lower while that of diesel fuel is slightly increased relative to the AEO 2010. LPG and natural gas prices (and hydrogen by extension) are sharply reduced in the AEO 2013 as a result of the sharp increases in shale gas production and proved reserves that have occurred since the AEO 2010 was released. The Monte Carlo simulations for each

pathway scenario are run in the same manner as described above. The primary difference is that the historical price variation probability distributions operate around the AEO 2013 prices for 2013-2032 rather than the AEO 2010 prices.

The results of the sensitivity analysis show that the effect of the AEO 2013 projected prices on each pathway scenario is largely determined by the level of hydrogen and/or natural gas consumption (see Figure 2.3). The HTG & FTS, LTG & FTS, and EH scenarios experience only marginal changes (<10%) to the mean NPV relative to the same scenarios under the AEO 2010 prices. None of these three scenarios consume external natural gas or hydrogen; furthermore, both FTS scenarios yield both gasoline and diesel fuel, allowing the lower price of the former to at least partially offset the higher price of the latter. Only the EH and MTG scenarios experience lower mean NPVs under the AEO 2013 prices than under the AEO 2010 prices. This reduction is particularly pronounced for the MTG scenario, which neither consumes hydrogen and/or natural gas nor yields a commodity that increases in value under the AEO 2013 relative to the AEO 2010.

All pathway scenarios benefit from higher gasoline and diesel prices, although EH does not benefit as much as the other scenarios due to the lower energy content (and consequent lower market value) of fuel ethanol relative to gasoline and diesel fuel. The impact of electricity prices depends primarily on whether the biorefinery is a net consumer or producer of electric power. Higher electricity prices result in a decrease to NPV for facilities that are net electricity consumers and an increase to NPV for those that are net electricity producers. Electricity production among net producers varies by pathway, ranging from 1 MW for the IHG & AAS scenario to 28.2 MW for the BFPH scenario. Facilities deriving a high proportion of their revenues from electricity see the greatest benefit from projected

increases in electricity prices. Similarly, natural gas prices have the most impact on the economic feasibility of biorefineries that depend heavily on an external supply of natural gas as either their source of hydrogen or heat and power. Natural gas could also impact LPG, which is a co-product of MTG and the market value of which is influenced by natural gas prices.

Two conclusions can be drawn from the results of this analysis. First, when historical energy commodity price variability is accounted for in projected prices over the next 20 years, both the mean and standard deviation for the NPV vary substantially across pathway scenarios. Second, when the effects of U.S. shale gas exploitation are considered in the form of updated EIA price projections (AEO 2013), the BFPH, JFPH, IHG & AAS, and DHG & AAS scenarios experience substantial (>10%) increases to mean NPV, while the MTG scenario experiences a substantial decrease to mean NPV.

These conclusions raise an important possible policy implication with regard to the RFS2. Lower natural gas prices both now and in the future improve the economic feasibility of those pathways that utilize natural gas as a pathway input, either directly or in the form of natural gas-derived hydrogen. While most of the pathways considered by this analysis can use biomass as a source of hydrogen rather than natural gas, such a substitution reduces pathway biofuel yields and decreases its economic feasibility as a result (93). Pathways such as gasification and FTS, FPH, and gasification and AAS therefore have a strong economic incentive to employ as much natural gas as is technically feasible, especially during early years of operation when natural gas prices are projected to be particularly low. While not all of the pathways considered by this analysis experience an increase to economic feasibility under the AEO 2013 price projections relative to those of the AEO 2010, economic

feasibility does benefit from reduced natural gas prices; however, the benefit isn't always sufficient to offset the other price changes in the AEO 2013. The two gasification and FTS pathway scenarios, two gasification and AAS scenarios, and EH scenario all are unable to generate a positive NPV under the AEO 2013 price projections even when natural gas is assumed to have no cost, indicating that they will struggle to be economically feasible as biorenewable pathways.

While reduced natural gas prices increase economic feasibility, the results of this analysis suggest that continued shale gas exploitation alone will be insufficient to make most of the pathways considered here economically feasible. Only two of the pathways considered (gasification and MTG and FPH) are found to have a greater than zero probability of achieving a positive NPV under the AEO 2013 projected prices, and even those probabilities might be too low to acquire the large amounts of capital from investors needed to achieve commercial-scale production. The RFS2 with its variable subsidy in the form of RINs can be expected to contribute to cellulosic biofuel economic feasibility as a result. While this subsidy has been worth nearly \$2/gal for other biofuel categories in the past (20), qualifying cellulosic biofuels must achieve a reduction to lifecycle GHG emissions of 60% relative to gasoline (95). As a fossil fuel, natural gas consumption increases pathway lifecycle emissions relative to biomass consumption. Consumption of too much natural gas, therefore, can disqualify a cellulosic biofuel pathway from the RFS2 and result in a net reduction to economic feasibility when RIN values are high. Furthermore, increased natural gas consumption can diminish the economic feasibility of pathways employing it in regions such as the European Union that impose a price on GHG emissions. While the use of natural gas as a hydrogen source doesn't necessarily disqualify pathways such as FPH on the basis of

lifecycle GHG emissions (35), further research is needed to quantify how much natural gas can be used as a pathway input before the pathway reduction to lifecycle GHG emissions falls below the 60% threshold for cellulosic biofuels.

2.5. Conclusion

This paper quantifies the 20-year net present value (NPV) for seven cellulosic biofuel pathways under eight different pathway scenarios using a discounted cash flow rate of return (DCFROR) spreadsheet. Input costs, output yields, and output prices are derived from techno-economic analyses (TEA) in the open literature and the Energy Information Administration's (EIA) Annual Energy Outlook (AEO) 2010. Uncertainty analyses for each pathway scenario are completed by developing fitted distributions based on historical price variations for diesel fuel, electricity, gasoline, LPG, and natural gas and applying them to the 20-year projected prices for each commodity in the AEO 2010.

The results of the uncertainty analysis show that the mean 20-year NPV and standard deviation calculation for each pathway scenario varies significantly, with the two fast pyrolysis and hydroprocessing (FPH) scenarios and single gasification and methanol-to-gasoline synthesis (MTG) scenario generating NPV probabilities in excess of zero. The lowest mean NPVs are generated by the four gasification scenarios and the single enzymatic hydrolysis (EH) and fermentation scenario. The two gasification and AAS scenarios produce the largest standard deviations of all pathway scenarios considered, indicating a high degree of NPV uncertainty for both.

A sensitivity analysis is also completed by replacing the AEO 2010 energy commodity prices with those from the AEO 2013, the latter accounting for the major increase

in U.S. natural gas production and proved reserves resulting from the recent exploitation of shale gas. This development is reflected in much lower natural gas and LPG prices in the AEO 2013 relative to the AEO 2010. Four of the eight pathway scenarios analyzed (BFPH, JFPH, IHG & AAS, and DHG & AAS) experience substantial increases to mean NPV when the AEO 2013 prices are considered. The gasification and MTG synthesis and EH and fermentation pathways experience decreases to mean NPV under the AEO 2013 prices, although the latter pathway experiences only a marginal decrease to mean NPV.



2.6. Supporting Materials

Figure S2.1. Methodology for the profitability analysis of biofuel scenario pathways based on commodity prices

Scenario	Pathway	Operating	Net	Natural gas	H ₂	Gasoline	Diesel	Energy
	_	hours/yr	electricity	consumption	consumption	output	fuel	efficiency
		_	required	(MT/day)	(MT/day)	(MGY)	output	$(\%)^{a}$
			(MW)	_	_		(MGY)	
S1 (29)	HTG &	7446	-13.8	0	0	12.9	25.8	53.2
	FTS							
S2 (29)	LTG &	7446	-16.4	0	0	10	20	42.7
	FTS							
S3 (69)	BFPH	7900	-28.2	0	49.1	28.7	28.7	59.5
$S_{4}(70)$	IEDH	7000	75	200 /	0	72.2	2.0	55 5
54 (70)	JFPH	/900	7.5	288.4	0	12.2	3.8	55.5
S5 (71)	IHG &	7900	-1	0	123.2	96.7	0	55.4
	AAS							
S6 (71)	DHG &	7900	14.2	155.8	149.4	1173	0	56.5
50(71)	AAS	1900	17.2	155.0	177.7	117.5	U	50.5
	1110							
S7 (72)	EH	8400	-25.8	0	0	35.6	0	45.3
<u>SQ (74)</u>	MTC	8400	0	0	0	42.5	0	12.6
58 (74)	MIG	8400	0	0	U	42.3	U	42.6

Table S2.1. Process scenario mass balances and annual operating hours. All scenarios assume 2000 metric tons per day (MTPD) of biomass input and \$83 per MT feedstock cost. (MGY – million gallons per year).

^a LHV basis

Scenario	Pathway	Total project	Fixed	Co-product	H ₂ cost	Other	Natural
		investment	operating	credit	(MM\$/yr)	variable	gas cost
		(MM\$)	cost	(MM\$/yr)		cost	(MM\$/yr)
			(MM\$/yr)	-		(MM\$/yr)	-
S1 (29)	HTG & FTS	657.4	15.5	6.8	0	12.6	0
S2 (29)	LTG (& FTS)	540.3	30.1	8.1	0	13.1	0
S3 (69)	BFPH	429.0	12.4	14.8	32.5	1.8	0
S4 (70)	JFPH	328.7	18.1	0	0	27.2	25.3
S5 (71)	IHG (& AAS)	627.1	36.7	0.5	77.9	88.1	0
S6 (71)	DHG (& AAS)	781.1	44.5	0	92.8	105.6	13.7
S7 (72)	EH	407.9	10.7	15.6	0	67.2	0
S8 (74)	MTG	216.5	15	13.4	0	2.1	0

Table S2.2. Capital and operating costs of process scenarios adjusted 2011 US dollar cost basis



Figure S2.2. Comparison of price histograms and probability density function fits.

Scenario	Pathway	Biomass feedstock type	Ultimate analysis	Moisture content
			(dry basis) (wt %)	(wt%)
		<u> </u>		22.0
SI (29)	HIG & FIS	Stover	Ash 6.0; C 47.3; H 5.1;	25.0
			N 0.8; CI 0; S 0.2; O	
			40.6	
S2 (29)	LTG & FTS	Stover	Ash 6.0; C 47.3; H 5.1;	25.0
			N 0.8; Cl 0; S 0.2; O	
			40.6	
S3 (69)	BFPH	Stover	Ash 6.0; C 47.3; H 5.1;	25.0
			N 0.8; Cl 0; S 0.2; O	
			40.6	
S4 (70)	JFPH	Hybrid poplar	N/A	50.0
S5 (71)	IHG & AAS	Hybrid poplar	N/A	50.0
S6 (71)	DHG &	Hybrid poplar	N/A	50.0
50(71)	AAS	ingonia popula		50.0
	11110			
S7 (72)	EH	Stover	Ash 6.0; C 47.3; H 5.1;	25.0
			N 0.8; Cl 0; S 0.2; O	
			40.6	
S8 (74)	MTG	Hybrid poplar	Ash 1.0; C 50.9; H 6.0;	50.0
			N 0.2; Cl 0; S 0.1; O	
			41.9	

Table S2.3. Compositions of biomass feedstocks



Figure S2.3. Projected annual prices from AEO 2010 and AEO 2013

4.1.3. Cellulosic Biofuel Pathway Descriptions

Figure S2.4 includes schematics of the pathways underlying the eight scenarios analyzed in this study. The fast pyrolysis, gasification with FTS, and gasification with AAS pathways each have two scenarios included in this study. The two fast pyrolysis scenarios are based on two studies of the same pathway; the primary differences between the two are the assumed capital cost factors and fuel yields. The two scenarios for the gasification with FTS and gasification with AAS pathways are based on pathway variations. All of the pathways analyzed convert lignocellulosic biomass into transportation fuels. Most of the pathway scenarios yield drop-in biofuels, although the EH and AAS scenarios yield ethanol. As noted earlier, some of these scenarios employ hydrogen derived from an external source. Brief descriptions of the pathways follow below.

4.1.4. Fast pyrolysis and Hydroprocessing

Fast pyrolysis is the rapid thermal decomposition of biomass in the absence of oxygen to liquid (bio-oil), solid (char), and gaseous (non-condensable gases, or NCG) coproducts. Bio-oil can be upgraded either catalytically via fluid catalytic cracking or with hydrogen via hydroprocessing to hydrocarbon-based gasoline and diesel fuel. Hydroprocessing is a two-stage upgrading process consisting of an initial low-severity hydrotreating step, which stabilizes and partially deoxygenates the bio-oil, and a subsequent higher-severity hydrocracking step, which fully deoxygenates and depolymerizes the hydrotreated bio-oil to monomeric hydrocarbons (see Figure S2.4-a). This hydrogen can be produced either by reforming a portion of the bio-oil produced via fast pyrolysis or reforming merchant natural gas; the former results in lower fuel yields while the latter increases the operating costs by adding an input (93).

4.1.5. Gasification and AAS

Gasification is the thermal decomposition of biomass at temperatures of up to 1500°C to a gaseous mixture of carbon monoxide (CO), hydrogen, methane, CO₂, and lesser amounts of light hydrocarbons. This product is known as synthesis gas, or syngas, and a number of routes exist for reacting clean syngas with catalysts or biocatalysts to produce liquid transportation fuels, including ethanol (71), gasoline (74), and diesel fuel (29). Although small amounts of char can be produced via biomass gasification, no bio-oil is produced and the primary product is syngas.

The AAS pathway employs multiple process steps to convert biomass to ethanol via gasification (see Figure S2.4-b) (71). The raw syngas is cleaned, at which point the CO and H_2 in the clean syngas are then reacted over a ZnO/CuO catalyst to yield methanol, which is distilled and in turn reacted with an iridium- and iodide-based catalysts to produce acetic acid. Finally, the acetic acid is reacted with hydrogen via hydrogenation to yield ethanol and water, which is distilled to produce fuel-grade ethanol.

4.1.6. Gasification and FTS

The gasification and FTS pathway (see Figure S2.4-c) resembles the AAS pathway in that the product syngas must be thoroughly cleaned to remove any impurities capable of poisoning the pathway catalysts. The CO and H_2 in the clean syngas are reacted over a cobalt, iron, or ruthenium catalyst to yield long-chain alkanes and hydrocarbon waxes. Both can be depolymerized to shorter hydrocarbon chains (so-called Fischer-Tropsch liquids) that can serve as refinery blendstock for the production of gasoline and diesel fuel.

4.1.7. Gasification and MTG Synthesis

The gasification and MTG synthesis pathway is similar to the gasification and AAS pathway in that the CO and H_2 in clean syngas are catalytically reacted to synthesize methanol. The pathways diverge after methanol synthesis, however, as the MTG pathway employs a methanol dehydration step to yield dimethyl ether (DME) (see Figure S2.4-d). The DME is reacted over a zeolite catalyst as a final step to yield a blend of aromatics and alkanes in the gasoline boiling range (74). A significant advantage of the MTG synthesis pathway over the FTS pathway is that the hydrocarbon synthesis products generally do not require depolymerization to short-chain alkanes, eliminating the need for hydrogen consumption via a hydrocracking step.

4.1.8. Enzymatic Hydrolysis and Fermentation

The production of ethanol via the enzymatic hydrolysis and fermentation pathway differs from the previous pathways in that it employs the biochemical platform rather than the thermochemical platform to both depolymerize lignocellulosic biomass and convert it to fuel. The fermentable monosaccharides glucose and xylose can be derived from cellulose and hemicellulose, respectively, and then fermented to ethanol. In addition to not being fermentable, however, lignin has anti-microbial properties that inhibit biochemical activity. The pathway therefore employs a feedstock pretreatment to separate the cellulose and hemicellulose from the lignin, which is combusted to provide electricity, and hydrolyze the hemicellulose to xylose. Following pretreatment, a system of cellulase enzymes is employed to hydrolyze the cellulose to glucose (96). The resulting monosaccharides are then fermented to ethanol, which is distilled to yield fuel ethanol (see Figure S2.4-e).



Figure S2.4. Schematics of pathways analyzed. a – fast pyrolysis and hydroprocessing; b – gasification and AAS; c – gasification and FTS; d – gasification and MTG synthesis; e – enzymatic hydrolysis and fermentation.

CHAPTER 3. A FRAMEWORK FOR DEFINING THE ECONOMIC FEASIBILITY OF CELLULOSIC BIOFUEL PATHWAYS

A paper submitted for publication in the international refereed journal *Biofuels* Tristan R. Brown⁴ and Mark M. Wright⁵

3.1. Abstract⁶

This paper incorporates pathway-specific financial assumptions into techno-economic analyses of cellulosic biofuel pathways under price uncertainty. Five cellulosic biofuel pathway scenarios are developed in a discounted cash flow rate of return spreadsheet to determine pathway-specific costs of debt. Cost of equity for the scenarios is calculated based on the financial characteristics of the U.S. biorenewable industrial sector. A 20-year net present value (NPV) and probability of default for each scenario are stochastically calculated. Mean NPVs vary from a low of -\$774 million to a high of -\$135 million. Probabilities of default range from a high of 100% to a low of 80.5%. A sensitivity analysis finds that the use of pathway-neutral financial assumptions overestimates NPV and underestimates probability of default.

⁴ Graduate student in Bioeconomy Institute, Iowa State University; primary researcher and author.

⁵ Assistant professor in Department of Mechanical Engineering, Iowa State University; corresponding author. ⁶ AEO = Annual Energy Outlook; CADS = cash available for debt service; CAPM = Capital Asset Pricing Model; DSCR = debt service coverage ratio; DCFROR = discounted cash flow rate of return; EBIT = earnings before interest and tax; EH = enzymatic hydrolysis and fermentation to ethanol; EIA = Energy Information Administration; EPA = Environmental Protection Agency; FPH = fast pyrolysis and hydroprocessing; FCI = fixed capital investment; FTS = Fischer-Tropsch synthesis; HTG = high-temperature gasification; IB = enzymatic hydrolysis and fermentation to isobutanol; ICR = interest coverage ratio; IRR = internal rate of return; LTG = low-temperature gasification; MMGPY = million gallons per year; MSFP = minimum fuel selling price; MT = metric ton; NPV = net present value; RFS2 = revised Renewable Fuel Standard; TEA = techno-economic analysis.

3.2. Introduction

One of the biggest challenges of the U.S. biofuels mandate, the revised Renewable Fuel Standard (RFS2), has been the inability of cellulosic biofuel pathways to produce more than a tiny fraction of the volumes required by the mandate to date. Created by Congress in 2007 to replace the obsolete original Renewable Fuel Standard (RFS1), the RFS2 establishes annual volumetric biofuel blending mandates that steadily increase from a total of 9 billion gallons on an ethanol-equivalent basis in 2008 to 36 billion gallons by 2022 (2). The majority of production in the later years of the mandate is to be accounted for by biofuels derived from lignocellulosic feedstocks, the mandated volume of which rapidly increases from zero in 2008 to 16 billion gallons (ethanol-eq.) in 2022. A number of cellulosic feedstocks are expected to contribute to this volume by 2022, with corn stover expected to be one of the primary contributors by weight (26). Investment in cellulosic biofuel capacity to date has been much lower than needed to produce the necessary volumes even in the RFS2's early years, however. As a result the Environmental Protection Agency (EPA), which oversees the RFS2, has revised the cellulosic biofuel volumes for 2010 through 2014 down by as much as 99% in response (see Figure 3.1). Even the revised volumes have proven to be overlyoptimistic in retrospect, however: zero gallons of cellulosic biofuel were blended in 2010 and 2011 (versus 6.5 million gallons and 6.0 million gallons revised, respectively), while only 0.02 million gallons and an estimated 0.82 million gallons were produced in 2012 and 2013 (versus 10.5 million gallons and 14.0 million gallons revised, respectively) (5). The EPA recently revised the 2014 cellulosic biofuel volume down from 1750 million gallons to 17 million gallons as capacity investment has continued to remain very low (24).



Figure 3.1. Originally-mandated, revised, and actual RFS2 cellulosic biofuel blending volumes, 2010-2014. 2013 actual volume estimated based on blending data through October 2013 (2,5,24).

The explanation for the lack of necessary capacity investment that is most commonly reported in the media is cellulosic biofuels' inability to compete in today's energy market, especially in light of its high capital costs. For example, multinational energy corporation Chevron determined in 2010 that investments in cellulosic hydrocarbon production generated only half of the return generated by conventional energy investments, causing it to scrap \$400 million in planned capacity investment (97). Multinational energy companies such as BP, Exxon Mobil, and Royal Dutch Shell have all made similar decisions in recent years for the same reason (98). These decisions have been supported by a 2011 report released by the National Research Council and published by the National Academy of Sciences (99), which relied upon techno-economic analyses (TEA) of three cellulosic biofuel pathways to arrive at

the conclusion that cellulosic biofuels will only be cost-competitive with petroleum-based fuels in an economic scenario in which high petroleum prices (> \$191/bbl), technological breakthroughs, and a high carbon price are the norm. However, this result is incongruous with a number of recent TEAs in the literature calculating cellulosic biofuel production costs to be below projected gasoline and diesel fuel prices (69,74,85,93,100,101), indicating that the latter calculations could be improved to better reflect the reality of cellulosic biofuel investment projects.

TEA is a methodology employed to estimate the production costs of energy pathways based on discounted cash flow rate of return (DCFROR) analysis. It is frequently used to estimate the production costs of biorenewable pathways that have yet to achieve commercialization. By calculating the production costs of various cellulosic biofuel pathways, TEA can be used to broadly estimate their economic feasibility for years in which petroleum price projections are also available (99). This use of TEA requires the accurate modeling of the specific technical and economic operating conditions of each pathway. A shortcoming of the current TEA methodology that is often employed to compare different pathways is that it treats technical conditions and economic or financial conditions differently: while the exact technical specifications of each pathway are carefully modeled, generic economic and financial assumptions are made that are identical across all pathways. For example, economic feasibility is commonly defined according to a financial standard, such as a minimum fuel selling price (MSFP) that is lower than the biofuel's market value (99) or a 20-year internal rate of return (IRR) that exceeds an arbitrary benchmark (102,103). Furthermore, the results of cellulosic biofuel pathway TEAs are highly uncertain due to the lack of commercialization in the sector. While some energy pathway TEAs attempt to

quantify this uncertainty by computing probability distributions via stochastic simulations (83,86,88), these analyses commonly employ statistical fit distributions that do not accurately reflect the range of potential values (87).

The use of a common set of economic and financial assumptions when completing TEAs of multiple pathways ignores their strong sensitivity to a pathway's technical specifications. Factors such as pathway yield, input requirements, and product portfolio have a large impact on the size and variability of the pathway's cash flows, which in turn are used to calculate economic feasibility metrics such as MFSP, IRR, and net present value (NPV). A recent comparison of eight different cellulosic biofuel pathway TEAs under price volatility and identical economic and financial assumptions found that both the mean and standard deviation of 20-year NPVs vary widely across pathways due to the influence of these factors, with the means ranging from a low of -\$590.2 million for enzymatic hydrolysis and fermentation to a high of \$274.8 million for fast pyrolysis and hydroprocessing (104). That comparison employed a 10% discount rate and 7.5% interest rate for each pathway; in reality, however, investors and creditors require both of these to be a function of NPV and cash flow variability, with higher costs of equity and/or debt required for those pathways that are expected to generate lower NPVs and/or higher cash flow variability. Given this wide variation in expected returns across cellulosic biofuel pathways, then, an accurate TEA comparison needs to treat each pathway's financial assumptions subjectively as differences between them will affect their economic feasibility determinations.

This analysis presents a framework for integrating TEAs of cellulosic biofuel pathways with uncertainty analysis to quantify the costs of debt and equity for each pathway. DCFROR spreadsheet models are created using cellulosic biofuel pathway data in the

techno-economic literature to quantify pathway economic feasibility. Pathway selection is based on those stover-based cellulosic biofuel pathways or their variants that are currently undergoing commercialization within the U.S. (27) and for which recent TEAs are available in the refereed literature. Uncertainty analysis is performed via Monte Carlo simulation to quantify pathway cash flows under commodity price uncertainty, which affects both pathway costs and revenue. The framework is then employed to quantify pathway-specific costs of debt and equity and to demonstrate the sensitivity of each pathway's economic feasibility determination to these financial costs.

3.3. Methodology

This analysis employs the following framework to incorporate pathway-specific financial costs into the economic feasibility quantification: (1) process and economic data of select stover-based cellulosic biofuel pathways are collected and adjusted; (2) historical data on monthly commodity price variation distributions are fit to probability distributions; (3) 20-year cash flows are deterministically simulated to calculate interest coverage ratios (ICR) for each pathway; (4) costs of debt and equity are calculated for each pathway as a function of their ICRs and industry betas, respectively; and (5) 20-year NPV distributions and 10-year probabilities of default are estimated for each pathway scenario as a function of stochastic cash flow simulation and cost of debt and equity calculations. Steps 1,2 and 5 are adopted from a previous comparative TEA of cellulosic biofuel pathways under uncertainty (104) while Steps 3 and 4 are unique to this analysis, as are the probability of default calculations. The methodology is illustrated in Figure 3.2.



Figure 3.2. Illustration of analysis methodology

Four different pathways are selected to develop five pathway scenarios with facility capacities of 2000 metric tons (MT)/day: (1) high-temperature gasification and Fischer-Tropsch synthesis (HTG & FTS); (2) low-temperature gasification and Fischer-Tropsch synthesis (LTG & FTS); (3) fast pyrolysis and hydroprocessing (FPH); (4) enzymatic hydrolysis and fermentation to ethanol (EH); and (5) enzymatic hydrolysis and fermentation to isobutanol (IB). The process data for the two gasification and FTS scenarios are based on an analysis by Swanson et al. (29). The FPH scenario is derived from an analysis by Brown et al. (69) and the EH scenario is based on an analysis by Kazi et al. (72). Finally, the IB scenario is developed from an analysis by Tao et al. (40). The original process conditions for each scenario are summarized in Table 3.1.

Scenario	Pathway	Operating hours/yr	Net electricity required (MW)	External H ₂ consumption (MT/day)	Gasoline- eq. output (MMGPY)	Diesel fuel output (MMGPY)
S1	HTG &	7446	-13.8	0	12.9	25.8
	FIS					
S2	LTG &	7446	-16.4	0	10	20
~-	FTS	,	1011	Ū.	10	_0
S 3	FPH	7900	-28.2	49.1	28.7	28.7
S4	EH	8400	-25.8	0	35.6	0
S5	IB	8400	-27.7	0	37.0	0

Table 3.1. Pathway scenario mass balances and annual operating hours. All scenarios assume 2000 MT (dry)/day of biomass input.

Multiple adjustments are made to the scenarios to account for major differences between the analyses on which the pathway scenarios are based. First, input and output prices are assumed to be the same for each pathway scenario. While these prices can be expected to vary across U.S. regions (32), this analysis makes this assumption for the sake of simplification and to remove regional price variation as a sensitivity factor. Second, all dollar figures are adjusted to a 2011 basis using the U.S. city average Consumer Price Index as the cost multiplier (76). Finally, the assumed feedstock cost has been adjusted to \$116.37/MT for all of the scenarios based on a techno-economic analysis of stover production and collection by Shah and Darr (2013) (105). Shah and Darr calculate a base case stover cost of \$117/MT under a diesel fuel price of \$3.50/gal and diesel fuel consumption of 3 gal/MT. The present analysis assumes an initial after-tax diesel price of \$3.29/gal and the assumed \$116.37/MT stover price reflects this reduced price. This analysis further assumes that farmer participation in stover harvesting is 50% (as opposed to the 30% value assumed by Shah and Darr) so as to ensure sufficient stover supply for a 2000 MT/day cellulosic biofuel facility. A 50% participation rate is assumed in other stover supply analyses (106,107) and has been

identified as the target rate for cellulosic biofuel producers (108). Other differences, such as capital cost calculation methodology and annual operating hours, remain unchanged from the original analyses. Table 3.2 summarizes the capital and operating costs for each of the pathway scenarios on an adjusted basis.

Scenario	Total	Fixed	Co-product	H ₂ cost	Other	Working
	project	operating	credit	(MM\$/yr)	variable	capital
	investment	cost	(MM\$/yr)		cost	% of
	(MM\$)	(MM\$/yr)			(MM\$/yr)	FCI
S1	657.4	15.5	6.8	0	12.6	30.4
\$2						30.8
52	540.3	30.1	8.1	0	13.1	50.0
S3	429.0	12.4	14.8	32.5	1.8	29.7
S4	407.9	10.7	15.6	0	67.2	31.8
S5	461.0	11.7	6.4	0	29.2	30.9

Table 3.2. Capital and operating costs of pathway scenarios adjusted 2011 US dollar cost

The Monte Carlo simulation employed by this analysis requires probability distributions to be selected for each of the commodity prices covered. These commodities are diesel fuel, electricity, gasoline, hydrogen (produced from natural gas via SMR), natural gas, and stover. They are selected due to (1) their use in cellulosic biorefineries and (2) the availability of sufficient historical monthly price data from which to identify probability distributions. The exception to the second criterion is stover, for which there currently exists no market from which prices can be identified. This analysis employs historical corn prices as a proxy from which to develop the price distribution for stover that is applied to the stover cost calculated by Shah and Darr (105). While the two feedstocks are used for different pathways, stover is a byproduct of corn production and stover yields are closely correlated to corn yields. The distribution selection methodology employed in this analysis is described in detail in a separate paper (104). An important difference between this analysis and the previous paper is that the former derives its probability distributions from the historical monthly price change over the previous month, rather than relative to the historical annual average price. This change places less certainty on average annual commodity price projections for the next 20 years and does not assume that prices will revert to the projected averages. The motivation behind this change is the underestimation of unforeseen abrupt changes and future consumption by previous long-term U.S. energy forecasts (109,110). For example, as late as 2010 the EIA projected a U.S. natural gas price for 2012 that ended up being 200% higher than the actual price for the year due to the unanticipated exploitation of domestic shale gas reserves and resulting surge in production (79,111). This analysis therefore considers price projection uncertainty rather than just price volatility, resulting in a broader range of projected price outcomes.

The best distributions for each commodity price are selected on the basis of the Anderson-Darling goodness-of-fit test results (91). The historical price changes are collected for the years 1993 to 2012 when available. Table 3.3 presents the distribution selected for each commodity price. To prevent unrealistic monthly price changes, each distribution is truncated to include those values falling within a 95% confidence interval (CI). Note that the hydrogen price is calculated as a function of the natural gas price given that half of global hydrogen production derives from natural gas reforming (112).

each commonly price						
Commodity	Distribution	Monthly price	Historical			
		change range	data years			
		(95% CI)	and source			
Electricity	Lognormal	-5.4%	2001-2012			
(industrial)		6.4%	(113)			
Diesel fuel	Logistic	-10.9%	1994-2012			
		11.1%	(114)			
Gasoline	Logistic	-14.6%	1993-2012			
		14.7%	(115)			
Natural gas (Henry	Logistic	-22.5%	1993-2012			
Hub)		23.0%	(111)			
Hydrogen ^a	Logistic		-			
Stover ^b	Logistic	-11.0%	1993-2012			
		11.0%	(116)			

Table 3.3. Probability distribution function for monthly change of each commodity price

^aCalculated as a function of natural gas price (80) ^bDistribution based on historical corn prices due to lack of a developed stover market

4.1.9. Cost of equity

This analysis calculates a cost of capital for each pathway scenario based on its respective pathway economics. A firm's cost of capital consists of both its cost of equity and cost of debt. The cost of equity is employed in the DCFROR model as the discount rate/IRR while the cost of debt is used to calculate interest expense within the model. The former is the rate of return investors require on an equity investment in a firm (117). Investors require a higher rate of return relative to the riskless rate from high-volatility investments than from low-volatility investments as compensation for the increased risk of the former. A publicly-traded firm's cost of equity can be calculated via the Capital Asset Pricing Model (CAPM) as a function of its historical stock returns relative to those of the market (also known as its "historical market beta"). The CAPM formula is:

$$E(R_i) = R_i + \beta_i [E(R_m) - R_i]$$
where $E(R_i)$ is the expected return on asset i, R_i is the risk-free rate, $E(R_m)$ is the expected return on a market portfolio, and β_i is the beta of asset i.

When the firm being analyzed is either privately-held or hypothetical, the latter being the case in this analysis, then there is no market beta from which to calculate E(R₁). An alternative approach, which is adopted here, uses the betas of the publicly-traded firms belonging to the same industrial sector as the firm being analyzed as a proxy beta (or "bottom-up beta") (118). The pathway scenarios presented in Table 3.1 are selected due to their employment (or the employment of a closely-related variant pathway) by existing or planned commercial-scale cellulosic biofuel facilities in the U.S. (27). Furthermore, several companies engaged in the production of either advanced biorenewable products, including biofuels (defined here as biorenewable pathways other than corn ethanol), or advanced biorenewable feedstocks have successfully staged initial public offerings (IPO) in recent years. This analysis calculates historical market betas for each of these publicly-traded companies based on their weekly stock returns from the date of their IPO to December 2, 2013 using the following formula to regress stock returns against market (defined here as the S&P 500 index) returns (117):

$$R_s = a + b R_m$$

where R_s is the stock return, R_m is the market return, *a* is the regression intercept, and *b* is the slope of the regression. Each return for the stocks and the overall S&P 500 is calculated as:

where j is the week for which the return is calculated and Adj. price is the return plus any dividends. The calculated stock betas and their respective standard errors are presented in Table 3.4. One outlier was removed from the quarterly data for a period when the

shareholder equity of Amyris fell to just above zero, resulting in a much higher-than-average debt/equity ratio for that quarter (6,015%).

Name	Ticker	Beta	Std. error	Market debt/equity	Effective
				ratio ^a	tax rate
Amyris	AMRS	1.68	0.42	82.1%	0%
Codexis	CDXS	2.23	0.29	0%	0%
Gevo	GEVO	2.14	0.44	36.5%	0%
FutureFuel	FF	1.10	0.19	0%	35.8%
KiOR	KIOR	0.96	0.50	42.1%	0%
Metabolix	MBLX	1.22	0.44	0%	0%
Syntroleum	SYNM	1.38	0.37	0%	0%
Solazyme	SZYM	1.58	0.30	25.7%	0%
Verenium	VRNM	0.95	0.36	390.0%	0%
Mean		1.47	0.37	64.0%	4.0%
Std. dev.		0.48	0.10	125.3%	12.0%
		So	urce: Yahoo	Finance	

Table 3.4. Advanced biorenewable companies historical market betas

^aQuarterly average since Q1 2011 or date of IPO (if more recent)

For practical purposes, the use of the biorenewable sector beta as a proxy beta for the hypothetical firms simulated in this analysis enables the calculation of non-arbitrary discount rates for each pathway scenario. Using the average beta from Table 3.4 within the CAPM formula yields a cost of equity of 10.0%, assuming a risk-free 20-year Treasury bond rate of 3.7% (119) and market risk premium of 4.3% (the geometric-mean risk premium for U.S. stocks over Treasury bonds from 1928 to 2010) (117). This is identical to the discount rate employed by a number of recent techno-economic analyses of cellulosic biofuel pathways (69,74,104,120–124). What this calculation doesn't account for, however, is the sensitivity of the advanced biorenewable sector's betas to its relative lack of development. Whereas the U.S. corn ethanol sector has an average facility capacity of 71.4 million gallons per year (MMPGY) as of November 2013 (125), the largest cellulosic biofuel facility currently operating is KiOR's 14 MMGPY facility in Columbus, Mississippi (126). Based on current

and proposed facility construction, the average U.S. cellulosic biofuel facility capacity is expected to be only 29.6 MMGPY by 2015 (27). The advanced biorenewable sector has seen more U.S. capacity investment than the cellulosic biofuel sector, although of the companies listed in Table 3.4 only Syntroleum (127) and Future Fuel (128) have existing advanced biofuel capacities in excess of 30 MMGPY (the former's capacity taking the form of a joint venture). The remaining companies must raise additional capital prior to investing in commercial-scale capacity and much of this can be expected to come in the form of debt, thereby increasing their market debt/equity ratios. The current state of the corn ethanol sector, which is more mature than the advanced biorenewable and cellulosic biofuel sectors and owns much more existing capacity, provides some support for this assumption. Table 3.5 lists the debt/equity ratios and effective tax rates for the four publicly-traded independent corn ethanol producers in the U.S., which have a combined biofuel capacity of roughly 1500 MMGPY. In addition to being characterized by a higher average market debt/equity ratio than the advanced biorenewable sector, the corn ethanol sector also has a higher effective tax rate. This is due to the fact that the majority of the companies listed in Table 3.4 have no taxable income and therefore also have an effective tax rate of 0%. As with Table 3.4, a single outlier was removed from Table 3.5 for a quarter when Pacific Ethanol's shareholder equity fell sharply and resulted in a quarterly market debt/equity ratio of 6,988%.

Name	Ticker	Market	Effective
		debt/equity ratio	tax rate
Biofuel Energy	BIOF	266.1%	0%
Green Plains	GPRE	137.3%	44.7%
Renewable Energy			
Pacific Ethanol	PEIX	320.3%	0%
REX American	REX	36.1%	22.6%
Resources			
Mean		171.6%	16.8%

Table 3.5. Publicly-traded independent corn ethanol companies

Table 3.5 continued					
Name Ticker Market Effective					
debt/equity ratio tax rate					
Std. dev 118.3% 21.4%					
Source: Yahoo Finance					

The present analysis assumes that commercial-scale facilities are operating for each cellulosic biofuel pathway scenario considered and therefore adjusts the average beta for the advanced biorenewable sector to account for this. Using the methodology described by Damodaran (2012) (117), an unlevered beta is first calculated via the following formula (117):

$$\beta_{\rm u} = \beta / [1 + (1 - t) \, {\rm x} \, ({\rm D}/{\rm E})]$$

where β_u is the unlevered beta for equity in a firm, β is the beta for equity in a firm, t is the firm's effective tax rate, and D/E is the firm's market debt/equity ratio. The unlevered beta disregards the effect of the debt/equity ratio and a value of 0.91 is calculated for the biorenewable sector. The unlevered beta is then levered using the desired debt/equity ratio; in this case a debt/equity ratio of 171.6% is employed to yield a levered beta of 2.48. When used in the CAPM formula with the 20-year Treasury yield of 3.7% and the historical market risk premium of 4.3%, this levered beta yields a cost of equity for the cellulosic biofuel sector of 14.3%, or 43% higher than the value commonly used in cellulosic biofuel techno-economic analyses.

4.1.10. Cost of debt

Cost of debt is calculated as a function of both a firm's debt/equity ratio and its cash flows, the latter determining whether the firm has sufficient cash to meet its interest payments and debt repayment. As with cost of equity, cost of debt is commonly calculated based on existing financial records – in this case, the interest rates paid either by the firm or

by other firms in the same industrial sector. Such a methodology is not suitable here due to the lack of cellulosic biofuel commercialization and a relative lack of debt in the advanced biorenewable sector (which is also due to a lack of commercialization to date). This analysis instead employs the interest coverage ratio (ICR) of each hypothetical stand-alone facility to determine cost of debt based on the relationship between the two in the broader market (see Table 3.6) (117,118).

market-capitanzation mins (117)						
Interest coverage ratio	Rating	Spread				
>12.5	AAA	0.50%				
9.5 - 12.5	AA	0.65%				
7.5 – 9.5	A+	0.85%				
6.0 – 7.5	А	1.00%				
4.5 - 6.0	A-	1.10%				
3.5 - 4.5	BBB	1.60%				
3.0 - 3.5	BB	3.35%				
2.5 - 3.0	B+	3.75%				
2.0 - 2.5	В	5.00%				
1.5 - 2.0	B-	5.25%				
1.25 – 1.5	CCC	8.00%				
0.8 - 1.25	CC	10.00%				
0.5 - 0.8	С	12.00%				
<0.5	D	15.00%				

Table 3.6. ICRs and ratings for low market-capitalization firms (117)

ICR is calculated by the following formula:

Interest coverage ratio = $\frac{\text{EBIT}}{\text{Interest expense}}$

where EBIT is earnings before interest and taxes. The spreadsheet model is used to deterministically calculate base case monthly ICRs for each pathway scenario under the riskfree rate and the calculated cost of equity, from which a mean ICR for the 10-year loan term is derived. The appropriate interest spread from Table 3.6 is then selected based on the respective ICR and added to the risk-free rate to calculate a loan interest rate for each pathway scenario. The spreadsheet model is changed to reflect this new interest rate and the process is repeated deterministically until the ICR and the interest rate converge. The converged deterministic ICR values are used to calculate mean stochastic ICRs for each scenario and the process is again repeated until the ICRs and interest rates converge. The final selected interest rates for each pathway scenario are presented in Table 3.7.

Scenario	Mean stochastic ICR	Interest rate
S1	-0.67	18.7%
S2	-0.99	18.7%
S3	0.49	18.7%
S4	-1.69	18.7%
S5	-1.05	18.7%

Table 3.7. Pathway scenario interest coverage ratios and interest rates

4.1.11. Monte Carlo simulation

The final step of this analysis stochastically calculates a 20-year NPV probability distribution and 20-year probability of default for each pathway scenario based on the values in Table 3.2, Table 3.3, and Table 3.7. This analysis adopts the approach presented by Esty (1999) (129) for simulating future commodity prices and all commodity price changes are assumed to follow a random walk with drift within the Monte Carlo simulation. The monthly changes for each commodity are based on the respective distributions and truncations presented in Table 3.3. Furthermore, each commodity price is assumed to drift, with the drift set to the value required for the mean of each monthly price distribution to match the respective annual price projections published by the Energy Information Administration for 2013-2032 (19). Crystal Ball® is used to generate 10,000 random monthly price changes per commodity and month, with the price in one month (P_t) equaling the previous month's price plus the random change (129):

 $P_t = P_{t-1}(1 + random change)$

The generated monthly price changes over the 20-year period covered by the DCFROR model are aggregated to produce 10,000 random price paths for each commodity. The price volatility of each commodity is based on its respective historical volatility. Furthermore, the historical price data is also analyzed with Crystal Ball[®] to identify correlations between different commodity prices via Spearman's correlation coefficient. A very strong correlation (>0.9) is identified for the prices of gasoline and diesel fuel (0.981) while a strong correlation (>0.5) is identified for the prices of diesel and corn (the latter on which the stover cost distribution is based). The Crystal Ball[®] simulations account for these correlations so as to prevent unrealistic price movements between commodities (e.g., a sustained sharp divergence between the prices of gasoline and diesel fuel). Finally, the monthly prices generated by Crystal Ball[®] are used to generate the monthly cash flows from which a 20-year NPV probability distribution is derived for each pathway scenario.

The 20-year probability of default is also calculated stochastically for each pathway scenario alongside the 20-year NPV. Probability of default is defined here as the probability that the mean minimum debt service coverage ratio (DSCR) of a pathway scenario is less than 1, as presented by Esty (1999) (129). The DSCR is a function of the scenario's cash available for debt service (CADS), where CADS is defined as EBIT plus depreciation minus income tax, and DSCR is defined as CADS divided by loan principal repayment. The DSCR calculation is performed each quarter (rather than each month) to reflect the corporate practice of making cash distributions on a quarterly basis (i.e., this analysis assumes that all cash remaining at the end of each quarter is distributed or expended). Any quarter in which the DSCR falls below 1.0 indicates that the pathway scenario facility has insufficient cash to make its loan repayments for the period. Finally, because each pathway scenario assumes

that facility output is only 50% of maximum during the first six months of production, this analysis increases the working capital value for each pathway scenario to the amount necessary to cover loan repayments until maximum output is achieved (see Table 3.2). This prevents the facility's initial underperformance from affecting the probability of default result.

3.4. **Results**

We calculate NPV and probability of default for five cellulosic biofuel pathway scenarios under electricity, diesel fuel, gasoline, natural gas, and stover price uncertainty. Four of the five pathway scenarios have very low probabilities of achieving NPVs that are greater than zero (see Figure 3.3). The HTG & FTS and LTG & FTS scenarios have probabilities of 0.3% and 0.04%, respectively. The EH and IB scenarios both achieve slightly higher probabilities of 0.9% and 1.2%, respectively. The FPH scenario has the highest probability of achieving a positive NPV at 27.2%. The FPH scenario also achieves the highest mean NPV at -\$134.7 million. The NPV standard deviations for all of the scenarios are large due to the wide range of commodity prices simulated by the Monte Carlo analysis, with the LTG & FTS scenario achieving the smallest standard deviation at \$161.4 million. The large standard deviations are expected due to the strong sensitivity of cellulosic biofuel pathways to the prices of the commodities considered in this analysis. The lowest maximum NPV is achieved by the HTG & FTS scenario, although even this exceeds \$545 million when the Monte Carlo simulation generates diesel fuel, gasoline, and electricity prices that match or exceed their respective historical highs. Similarly, however, very low NPVs are reported

Scenario 1: HTG & FTS Scenario 2: LTG & FTS 0.14 0.14 (\$MM) (\$MM) -742.9 Mean -749.4 Mean 0.12 0.12 -780.1Median Median -761.6 $SD(\sigma)$ 203.1 $\mathrm{SD}\left(\sigma\right)$ 161.4 0.10 0.10 Probability 90'0 80'0 Probability 90'0 0.04 0.04 0.02 0.02 0.00 0.00 _1500. -1000. _2000. _500. 2000. _1500. _1000. 500. 1000. 1000. 500. 0. 500. 0 Net Present Value (MM) Net Present Value (MM) Scenario 3: FPH Scenario 4: EH & Fermentation 0.14 0.14 (\$MM) -134.7 (\$MM) Mean Mean -773.9 0.12 0.12 -815.9 Median -153.6 Median $SD(\sigma)$ 272.5 $SD(\sigma)$ 242.8 0.10 0.10 Probability 90'0 80'0 Probability 90.0 0.04 0.04 0.02 0.02 _1500. 0.00 2000. 0.00 _1500. _1000. 1000. 2000. _1000. 1000. 500. _500. 500. 0. 500. 0 Net Present Value (MM) Net Present Value (MM) Scenario 5: Isobutanol 0.14 (\$MM) Mean -678.9 -716.2 0.12 Median $SD(\sigma)$ 228.1 0.10 Probability 90.0 0.04 0.02 0.00 2000. _1500. 1000. _1000. _500. 0 500. Net Present Value (MM)

for all of the pathway scenarios when the Monte Carlo simulation generates high input costs and low output prices.

Figure 3.3. 20-year NPV results for five cellulosic biofuel pathway scenarios

The probability of default calculations present a slightly more optimistic result for the economic feasibility of the five pathway scenarios. While the HTG & FTS, LTG & FTS, and EH scenarios have a less than 1% probability of achieving positive NPVs, their probabilities of defaulting during the 10-year loan term are all 100%. The probability of default for the IB scenario is also 100%. The FPH scenario achieves the lowest probability of default, although it is still high at 80.5%. While the range of commodity prices generated by the Monte Carlo simulation impacts the probability of default calculations, the latter are also sensitive to the pathway scenario's product portfolio diversity. The FPH scenario is unique in that it yields equal volumes of gasoline and diesel fuel and the largest amount of electricity. The EH and IB scenarios stand in sharp contrast to the FPH scenario, generating only a single liquid fuel product in the form of ethanol and butanol, respectively, and utilizing a single feedstock in the form of stover (the three thermochemical scenarios all utilize both stover and natural gas as inputs). The probability of default results indirectly affect the NPV results in turn, as the related ICR determines the cost of debt of each pathway scenario. In other words, a high initial probability of default results in a low initial ICR, which increases the cost of debt and thereby increases the probability of default still further. NPV is therefore an important measure of economic feasibility but not the only one that should be considered in TEAs, as a cellulosic biofuel pathway with a high NPV and high probability of default is less likely to contribute to the RFS2 than a less profitable pathway that is able to avoid default via product portfolio diversity.

The results of this analysis suggest that government subsidization of cellulosic biofuels production and/or consumption will be required if corn stover is to be a viable feedstock under the RFS2. None of the five pathway scenarios considered achieve a positive

average 20-year NPV. Of equal importance are the very high probabilities of default calculated for the HTG & FTS, LTG & FTS, EH, and IB pathway scenarios during their first decade of operations, which indicate that they will be unlikely to contribute to the RFS2 for even a majority of their 20-year lifespans. In reality it will be difficult for pathways presenting investors with the near-certainty of default to acquire the necessary capital investment to begin production in the first place. While government subsidies have the potential to reduce the likelihood of default for the cellulosic biofuel pathways considered and improve the ability of stover to contribute to the RFS2 as a cellulosic feedstock, most of the stover pathways considered by this analysis are unlikely to achieve economic feasibility in their absence.

3.5. Sensitivity analysis

Two sensitivity analyses are completed to determine the influence of the analysis assumptions on its results. The first sensitivity analysis is performed via Crystal Ball's Tornado Chart[®] tool to quantify the sensitivity of the ICR of each pathway scenario to each commodity. A testing range between the 20th and 80th percentile of each commodity price distribution in the first month of the analysis (sensitivity steadily declines which each subsequent month) is employed to calculate a range of ICR values. The results of the sensitivity analysis are presented in Table 3.8.

	Table 5.8. Sensitivity of Tex to commodity prices					
	S 1	S 2	S 3	S 4	S5	
Commodity		ICR range				
Diesel fuel	0.075	0.071	0.190	0.000	0.000	
Electricity	0.004	0.006	0.019	0.013	0.012	
Gasoline	0.047	0.043	0.241	0.230	0.181	
Natural gas	0.001	0.001	0.067	0.000	0.000	
Stover	0.069	0.084	0.168	0.126	0.105	

Table 3.8. Sensitivity of ICR to commodity prices

The ICRs of all five of the pathway scenarios are most sensitive to changes in the price of the pathway's primary transportation fuel output. The two gasification and FTS scenarios produce twice as much diesel fuel as gasoline by volume and their respective ICRs are much more sensitive to changes in the price of the former than of the latter as a result. The HTG & FTS scenario is more sensitive to the price of diesel fuel than to the other commodities. The LTG & FTS scenario is most sensitive to the stover cost, however, due to its lower fuel yield relative to the HTG & FTS scenario. The EH and IB scenarios produce no diesel fuel (or a substitute) and are most sensitive to the price of gasoline as a result. While the FPH scenario produces equal volumes of gasoline and diesel fuel it is more sensitive to the price of the former due to the broader range of gasoline prices employed in the Monte Carlo simulation (see Table 3.3). The ICRs of the FPH, EH, and IB scenarios are all very sensitive to the stover cost, although this is not as important as the transportation fuel price due to the relatively high fuel yields attained by each of the scenarios. All of the pathway scenario ICRs are only weakly sensitive to the prices of electricity, indicating that there is both relatively value to be attained via electricity generation and low electricity price uncertainty. Only the FPH pathway scenario is sensitive to the price of natural gas, and even this is weak relative to the scenario's sensitivity to transportation fuel prices and the cost of stover. Of the commodities considered by this analysis, then, only the primary feedstock and products can be expected to have a substantial impact on pathway ICR.

The second sensitivity analysis employs Monte Carlo simulation to calculate 20-year NPVs and 10-year probabilities of default for each pathway scenario using the identical financial assumptions of a 100% debt/equity ratio, 10% IRR, 7.5% interest rate, and working

capital equal to 15% of fixed capital investment (FCI). Employment of these identical financial assumptions generates NPVs and probabilities of default that are in some cases substantially higher and lower, respectively, than under the pathway-specific assumptions based on unique pathway scenario cash flows (see Table 3.9). The magnitude of this difference varies by pathway scenario, however. The probabilities of default for the HTG & FTS, LTG & FTS, EH, and IB scenarios are only marginally lower (if at all) under the identical financial assumptions, due to the negative returns under both sets of assumptions. The 20-year NPVs of the four scenarios increase by as much as \$282 million but remain too low to generate low default probabilities. The exception to this is the FPH scenario, for which a 10-year default probability of 45.3% is calculated under the identical financial assumptions. The reason for the pathway scenario's sharp reduction to probability of default is the corresponding increase to a positive NPV under the identical financial assumptions. Similar results could be expected for the other pathways were their 20-year NPVs not so low to begin with. The results of the second sensitivity analysis indicate that the use of identical financial assumptions in TEAs of cellulosic biofuel pathways generate results that overstate their economic feasibility, although the magnitude of this overstatement varies by pathway.

	Pathway-speci	fic financial	Identical financia	l assumptions
	assumptions			
Scenario	Mean NPV (\$MM)	Default prob.	Mean NPV (\$MM)	Default prob.
S 1	-\$749.4	100%	-\$467.0	98.5%
S2	-\$742.9	100%	-\$543.2	100%
S 3	-\$134.7	80.5%	\$133.1	45.3%
S4	-\$773.9	100%	-\$687.0	100%
S5	-\$679.0	100%	-\$499.0	99.4%

Table 3.9. Sensitivity of results to financial assumptions

3.6. Conclusion

This paper quantifies the 20-year net present value (NPV) and 10-year probability of default for five stover-based cellulosic biofuel pathway scenarios using a discounted cash flow rate of return spreadsheet. Unlike previous techno-economic analyses (TEA), this paper calculates pathway-specific costs of debt and equity for each pathway scenario based on industry cash flows and scenarios' respective cash flows. Feedstock requirements, output yields, product portfolios, and capital costs are derived from recent TEAs in the refereed literature. Input costs and output prices for the five energy commodities covered are calculated stochastically via Monte Carlo simulation as a random walk with drift, with the drift based on the 20-year price projections in the U.S. Energy Information Administration's Annual Energy Outlook. Finally, two sensitivity analyses are employed to quantify the sensitivity of the interest coverage ratio (ICR) to the commodity prices and of the NPV and probability of default calculations to the type of financial assumptions made.

This paper calculates negative mean NPVs for all of the pathway scenarios. Furthermore, it calculates very high (100%) probabilities of default for all of the scenarios except the fast pyrolysis and hydroprocessing scenario. The wide range of stochastic price paths simulated by the analysis results in large 20-year NPV standard deviations for all of the pathway scenarios. The first sensitivity analysis identifies the ICR as being most sensitive to the transportation fuel prices and feedstock costs, although the magnitude and ranking of this sensitivity varies by pathway. Finally, the second sensitivity analysis finds that the use of arbitrary financial assumptions across all scenarios results in the overestimation of 20-year NPVs and underestimation of 10-year default probabilities.

The authors are grateful for the financial support of the Biobased Industry Center under Grant number 2703664, and the Iowa National Science Foundation EPSCoR project grant EPS-1101284.

CHAPTER 4. QUANTIFYING THE ECONOMIC COMPETITIVENESS OF INITIAL U.S. CELLULOSIC BIOFUEL PATHWAYS

A paper to be submitted for publication in an international refereed journal Tristan R. Brown¹ and Mark M. Wright²

4.2. Abstract

This paper calculates 20-year net present values (NPV) and 10-year probabilities of default for six cellulosic biofuel pathways under eight location scenarios. Each location scenario resembles one of the nine commercial-scale cellulosic biofuel projects planned in the U.S. at the end of 2012 and accounts for location-specific factors such as capital costs, feedstock type, feedstock costs, energy commodity prices, and state corporate income tax rates. Negative 20-year NPVs and low probabilities of positive NPVs are calculated for seven of the eight location scenarios. Very high (>98%) probabilities of default are also calculated for seven of the eight location scenarios, while every scenario is calculated to have a probability of default above 50% over 10 years.

4.3. Introduction

The Energy Independence and Security Act of 2007 (EISA) mandates the consumption of increasing volumes of biofuels derived from lignocellulosic biomass ("cellulosic biofuels"). The mandate, as part of the larger revised Renewable Fuel Standard (RFS2), requires this biofuel category to become the largest biofuel by volume in 2022 (2). Initial optimism regarding the economic and technical feasibility of cellulosic biofuel pathways quickly flagged as production routinely failed to make its mandated appearance

¹ Graduate student in Bioeconomy Institute, Iowa State University; primary researcher and author.

² Assistant professor in Department of Mechanical Engineering, Iowa State University; corresponding author.

(130,131). The Environmental Protection Agency (EPA) oversees the RFS2 and has greatly reduced the mandate's original volumes for each of the last four years due to a lack of cellulosic biofuel capacity (2). Despite these modifications, the annual adjusted mandates have consistently exceeded the supply of cellulosic biofuels. In 2010 and 2011 the EPA adjusted the original 100 million gallon and 250 million gallon cellulosic biofuel mandates to 6.5 million gallons and 6.6 million gallons, respectively. The 2012 mandate called for 500 million gallons but was reduced by the EPA to 10.5 million gallons. Similarly, the original 2013 mandate of 1000 million gallons of cellulosic biofuel consumption was later reduced to 14 million gallons.

Despite these adjustments by the EPA, actual cellulosic biofuel consumption has continued to fall far short of the adjusted volumes due to a lack of production. No consumption occurred in 2010 and 2011 (11) while the actual volumes for 2012 and 2013 were 99% and 97% below the adjusted mandates, respectively (5,12). In 2013 a U.S. appellate court ruled that the methodology on which the EPA was basing its annual adjusted volumes was weighted toward overproduction and therefore flawed (13). The EPA's position is that skewing its projections toward overproduction improves the economic feasibility of cellulosic biofuel pathways since the value of the mandate's flexible subsidy component (Renewable Identification Numbers, or RINs) increases when actual production falls short of the volumetric mandate (132). This approach has prompted multiple lawsuits from the petroleum industry, which bears the cost of the flexible subsidy, against the EPA, as well as Congressional proposals to eliminate the RFS2's so-called "phantom fuels" mandate (133).

The debate over the EPA's projections has been driven by the fact that the RFS2 has not incentivized nearly enough capacity investment in cellulosic biofuel pathways to meet the

mandated consumption volumes. The RFS2 mandates the annual purchase of cellulosic biofuel by "obligated blenders", generally petroleum refiners. It encourages production of the corresponding biofuel via RINs, which are a flexible subsidy payment from obligated blenders to cellulosic biofuel producers and non-obligated blenders for every gallon of cellulosic biofuel that is blended with gasoline or diesel fuel for retail. In order to prevent the subsidization of windfall profits, the core RIN value operates as a function of the cellulosic biofuel's production cost and the market price of petroleum; the RIN value increases when the biofuel's production cost increases and decreases when petroleum's market value increases, so long as total production does not exceed the volumetric mandate (2,20). This flexible subsidy has worked well for the other biofuel categories under the RFS2 due to the existence of significant capacity at the time of the program's implementation. It has had little success encouraging cellulosic biofuel production and capacity investment to date, however, because the core RIN value can only be calculated when production exists and production costs are well-known. Since no cellulosic biofuel production occurred in 2010 and 2011 and only minimal volumes were produced in 2012 and 2013, however, neither cellulosic biofuel production costs nor RIN values are well-known.

The lack of an established RIN price has discouraged investment in cellulosic biofuel capacity by creating uncertainty as to the ability of cellulosic biofuel pathways to compete with fossil fuel pathways, particularly petroleum refining. Efforts to estimate future cellulosic biofuel RIN values have been hampered by the diversity of pathways qualifying for the cellulosic biofuels mandate, with each possessing a unique set of technical and economic specifications. One comparative techno-economic analysis (TEA) of the three pathways closest to commercialization calculates minimum fuel selling prices (MFSP) varying from a

low of \$2/gallons gasoline-eq. (gge) to a high of \$5.50/gge (29–31,93). A subsequent attempt by the National Research Council (NRC) to calculate the necessary total cost of cellulosic biofuel subsidization under the RFS2 as the difference between these results and assumed transportation fuel prices presents a range of \$31 billion due to this uncertainty (99). At the beginning of 2013 there were at least six different pathways expected to achieve commercialscale (>20 million gallons per year) production by the beginning of 2015 (27) as opposed to the three considered by the National Research Council, suggesting that the actual subsidy cost range could be larger still.

Further complicating efforts to quantify future cellulosic biofuel RIN prices is the expected regional variation in production costs for individual pathways. A number of variables to which the economic competitiveness of cellulosic biofuel pathways is sensitive, such as capital costs, feedstock costs, and product market prices, vary substantially across regions. A recent analysis calculates that the 20-year net present value (NPV) of a 2000 metric ton (MT)/day fast pyrolysis and hydroprocessing (FPH) facility varies from -\$79.5 MM to \$165.5 MM depending on the U.S. state in which it is located (32). Potential investors and creditors are unlikely to provide the capital necessary for construction of a commercialscale cellulosic biofuel facility unless an adequate return on capital is likely to be achieved and this will not be possible in all potential locations absent the provision of government support. Many TEAs of cellulosic biofuel pathways published in the refereed literature limit capital cost calculations to a U.S. Gulf Coast basis due to the heavy concentration of fossil fuel-related chemical engineering projects in the region. Aspen Capital Cost Estimator® and the CHEMCAD[®] software suite are two programs that are commonly employed for the techno-economic modeling and capital cost estimation of cellulosic biofuel pathways

(29,72,74,85,93,134). Sufficient concentrations of lignocellulosic biomass to supply commercial-scale facilities are available in U.S. regions outside of the Gulf Coast such as the Midwest (135), however, and capital cost factors vary across these regions due to differences in infrastructure, equipment and materials availability, labor costs, weather, and seismic activity (136).

Region-sensitive factors that also influence the calculated economic competitiveness of cellulosic biofuel facilities in addition to capital costs are feedstock costs, transportation fuel prices (excluding state fuel excise taxes), and state corporate income tax rates (32). Feedstock type is region-specific due to differences in growing conditions, ecosystems, and agricultural practices across regions. For example, the U.S. Midwest is expected to be a major future supplier of corn stover to cellulosic biofuel producers due to the large amounts of corn already grown in the region. The U.S. Southeast, on the other hand, is already a major provider of softwoods to the paper and pulp industry and is expected to contribute forestry feedstock to the cellulosic biofuel sector (26). Furthermore, differences in the level of market development for each feedstock affect the level of uncertainty regarding the price of each. While detailed market prices are widely available for softwoods in major timber-producing states, the absence of stover collection on a commercial-scale at the time of writing makes estimated stover costs and market prices highly uncertain. The assumed feedstock cost for a Midwest biorefinery utilizing stover is much less certain than that for a Gulf Coast counterpart utilizing softwood, other things being equal.

Finally, commodity price volatility and uncertainty have both been shown to impact the 20-year NPVs of cellulosic biofuel pathways (104). Commodity prices, especially energy prices, exhibit high volatility and in turn increase the cash flow volatility of the facilities that

employ them as inputs (natural gas) and/or outputs (transportation fuels). High cash flow volatility increases the probability that a producer's revenue will be temporarily insufficient to meet its periodic debt payments, in the event of which a default occurs and continued future operations at the facility are unlikely. Accounting for this anticipated future price volatility in TEAs of cellulosic biofuel pathways is important as it generates a range of possible financial results with corresponding probabilities rather than a single result that is erroneously believed to be known with complete certainty (104,137). Accounting for uncertainty in price projections is important for the same reason, as projections are not known with complete certainty even when price volatility is not considered.

While U.S. cellulosic biofuel commercialization to date has fallen well below the volume necessary to meet the initial mandates of the RFS2, nine facilities with a combined 323 million gallons per year (MMGY) of capacity on an ethanol-equivalent basis were expected at the beginning of 2013 to be in operation within two years (27). The announced locations of these facilities are in eight different states, with two in Iowa and one apiece in Louisiana, Kansas, Michigan, Mississippi, North Carolina, Oregon, and Tennessee. Efforts such as that by the NRC to calculate the financial returns of these facilities and subsequent subsidization costs of the RFS2 on the basis of Gulf Coast capital costs, national commodity prices, and identical financial assumptions do not reflect this diversity of cellulosic biofuel facility pathways and locations. Accounting for this regional diversity can be expected to provide more realistic TEA results of the pathway facilities modeled and a more accurate estimate of the RFS2's short-term subsidization costs than would be generated without its consideration.

The objective of this analysis is to quantify the economic competiveness of the six pathways to be employed by the nine facilities expected to be in operation in 2015, with each facility serving as an individual location scenario, within their specific state locations and under price uncertainty. Eight location scenarios are developed (two of the nine facilities are expected to employ the same pathway in the same location and only one is considered here as a result) and each scenario is based on a recent cellulosic biofuel pathway TEA from the open literature for the appropriate scenario (although in two cases the TEA is for a similar but not identical pathway).

4.3.1. Methodology

4.3.2. Scenario selection

Eight location scenarios are developed to represent the nine commercial-scale cellulosic biofuel facilities (defined here as those with annual biofuel capacities exceeding 20 MMGY) that are expected to be in operation in the U.S. by 2015. Each scenario broadly resembles one of the nine commercial-scale cellulosic biofuel facilities that is expected to be operating by the beginning of 2015 (27). While the pathways considered are selected based on their U.S. commercialization status, the pathways for which TEAs are available in the open literature are not always identical to those that are being commercialized. For example, Scenario 3 considers the production of ethanol from lignocellulosic biomass via gasification, methanol synthesis to acetic acid, and acetic acid hydrogenation processes. While Zeachem is building a commercial-scale facility in Oregon that will employ a similar process for the conversion of acetic acid to ethanol, the company's acetic acid is produced via biomass dilute acid hydrolysis and fermentation of the resulting substrate (138). Since there are no TEAs of that specific biochemical pathway for the production of acetic acid in the literature, Scenario 3 considers the related thermochemical pathway (71). Similarly, Scenario 1 is based on a TEA of a mild catalytic pyrolysis process that is similar but not identical to the pathway employed by KiOR, which employs a proprietary catalytic pyrolysis process. Figure 4.1 presents the steps employed by this analysis for developing each location scenario. Table 4.1 presents the pathway and relevant operating assumptions employed for each location scenario (ethanol output is presented on a gasoline-equivalent basis).



Figure 4.1. Methodology employed by analysis. Note that corporate income taxes are not included in flowchart for simplicity.

Location Pathway Gasoline-Diesel fuel Net electricity Literature Location Assumed scenario eq. output output output (MW) feedstock feedstock type for cost (MMGY) (MMGY) 33.7 S1 (120) Mild catalytic MS 26.3 12.3 Hybrid Shortpyrolysis & rotation poplar hydroprocessing to woody crops hydrocarbons S2 (139) Consolidated MI 23.1 26.8 0 Forest Corn stover bioprocessing to residue ethanol S3 (71) OR 117.3 -14.2 Wood Direct-heat 0 Forest gasification & acetic residue chips acid synthesis to ethanol Enzymatic hydrolysis KS 35.6 25.8 S4 (30) 0 Corn stover Corn stover to ethanol Enzymatic hydrolysis S5 (30) IA 35.6 0 25.8 Corn stover Corn stover to ethanol Enzymatic hydrolysis NC Corn stover S6 (30) 35.6 0 25.8 Corn stover to ethanol 13.8 S7 (29) High-temp TN 12.9 25.8 Forest Corn stover gasification & F-T residue synthesis to hydrocarbons S8 (74) LA 42.5 0 Poplar Gasification & MTG 0 Forest synthesis to residue wood hydrocarbons

Table 4.1. Location scenario details.

4.3.3. Feedstock cost

The lignocellulosic feedstock assumption for each location scenario is made within the constraints of data availability in the open literature. The TEAs used to develop each location scenario are based on a single lignocellulosic feedstock each. Frequently the assumed type of lignocellulosic feedstock in a pathway TEA is not always the same as that being employed by a commercial-scale counterpart biorefinery in practice. Rather than exclude an otherwise-suitable pathway from this analysis whenever such a difference occurs, it is instead assumed that the pathway TEA's feedstock is employed by the relevant location scenario from a processing standpoint in terms of calculating capital costs and process yields. However, the feedstock price most likely to be encountered in practice by the relevant pathway based on current commercialization trends is used in the discounted cash flow rate of return (DCFROR) analysis. Such an assumption is made for Scenarios 2 and 7, which utilize pathway TEAs in which stover is the pathway feedstock but assume that the stover is made available at the cost of forest residue feedstock (see Table 4.1). It should be noted that feedstock type can be expected to influence compositions and yields of the final products, although at this time there is not a clear correlation between individual lignocellulosic feedstocks and pathway yields; see Elliott et al. 2009 (140) for a thermochemical pathway example. Finally, all of the location scenarios considered by this analysis are within states that have been previously identified as having sufficient feedstock to supply a biorefinery with feedstock demand of up to 2000 MT/day (32).

The NRC's calculation of the costs of cellulosic biofuel pathways employs the <u>Bio</u>fuel <u>Break</u>even Model (BioBreak) to determine lignocellulosic biomass producers' willing to accept (WTA) price for multiple feedstocks and U.S. regions (99). The BioBreak

model calculates the WTA prices on the assumption of a \$120/bbl crude petroleum price (2011 dollars), which is close to the Energy Information Administration's (EIA) 2013 Annual Energy Outlook (AEO) average 20-year price projection for West Texas Intermediate (WTI) petroleum of \$109.45/bbl (19). This analysis utilizes the BioBreak model's WTA price (adjusted to a 2011 cost basis) results under the \$120/bbl petroleum price assumption as the feedstock prices for each of the respective location scenarios (see Table 4.2). Since Scenarios 2, 7, and 8 assume the use of feedstocks that are not utilized by their real-life counterparts, however, the feedstock costs most likely to be incurred by the latter based on the BioBreak results are employed in this analysis.

Location	Electricity	Natural gas	Gasoline	Diesel fuel	Feedstock	State corp.	Location	Adj. TPI
scenario	price, 2013-	price, 2013-	price ex-tax,	price ex-tax,	cost	tax rate	capital cost	(MM
and	2032	2032	2013-2032	2013-2032	(\$/dry MT)	(%)	factor	2011
source	(\$/kWh)	(\$/MMBTU)	(\$/gal)	(\$/gal)				US\$)
S1 (120)	0.055-0.056	4.74-7.32	3.05-3.48	3.16-3.98	\$93	5.0	0.99	452.0
S2 (139)	0.065-0.071	5.24-7.63	3.06-3.51	3.16-4.03	\$93	6.0	1.28	458.2
S3 (71)	0.074-0.076	5.30-7.98	3.22-3.82	3.19-4.33	\$106	7.0	1.28	1,001.7
S4 (30)	0.058-0.064	4.45-7.05	3.00-3.48	3.13-4.00	\$110	7.0	1.06	431.8
S5 (30)	0.058-0.064	4.45-7.05	3.00-3.48	3.13-4.00	\$110	12.0	1.15	470.2
S6 (30)	0.063-0.068	5.30-7.64	3.05-3.56	3.17-4.08	\$119	6.9	0.94	383.8
S7 (29)	0.055-0.056	4.74-7.32	3.05-3.48	3.16-3.98	\$93	6.5	1.01	665.1
S8 (74)	0.051-0.068	2.94-5.37	3.01-3.46	3.13-3.96	\$93	8.0	1.00	216.5

Table 4.2. Location scenario market conditions. Price ranges based on projections from 2013 to 2032.

4.3.4. Location capital cost factor

Facility location also impacts its total capital cost. Capital cost estimates generated by process engineering software are commonly reported in the open literature on a U.S. Gulf Coast basis, as chemical engineering equipment is frequently sourced in that region due to its high concentration of petroleum refineries and transport hubs. Constructing the same biorefinery in another region incurs region-specific factors that affect its capital costs, including differences in equipment transport, equipment availability, local weather, seismic activity, and wages (136). A previous study reviews the various regional construction cost indices that have been created for the purpose of calculating capital costs for a given location and baseline (32). These include the Department of Defense (DOD) United Facilities Criteria (136), the Richardson International Construction Factors ManualTM (141), and the ENR 20-city Construction Cost Index (142).

This analysis utilizes area cost factors published by the DOD due to their inclusion of hundreds of U.S. locations, both urban and rural (136). The Richardson International Construction Factors ManualTM and ENR 20-city Construction Cost Index are limited to urban costs that are unlikely to be encountered by bioenergy facilities due to the high costs that would be incurred by transporting biomass feedstock from rural production areas (75). The DOD's area cost factors are based on local costs for a basket of eight labor activities, 17 construction materials, and four equipment items. State average factors are attained from the index for each location scenario and adjusted to place them on the same basis by using the average state area cost factor for Louisiana as unity. The adjusted average area cost factors for each state are then applied to the unadjusted total project investment (TPI) of each relevant location scenario to calculate an adjusted TPI (see Table 4.1). This adjusted TPI is

employed in the discounted cash flow rate of return (DCFROR) analysis as the biorefinery capital cost.

4.3.5. State corporate income tax rate

While the U.S. federal corporate income tax rate for corporations is uniform with a top bracket rate of 35%, business entities must also pay corporate income tax at the state level. The top rate varies by state, from 12% in Iowa to zero in South Dakota (143). Each location scenario in this analysis accounts for both the 35% federal rate as well as the respective state corporate income tax rate as part of the DCFROR analysis. This analysis does not account for unique incentives, such as tax credits and low-interest loans that some states provide to specific enterprises operating within their borders.

4.3.6. Energy commodity prices

The market prices of gasoline and diesel fuel vary on a regional basis even after statelevel differences in fuel excise taxes are accounted for (92,144). Similar price differences are witnessed for the market prices of electricity and natural gas as well (see Table 4.2). This analysis assumes that each facility simulated by the respective location scenario purchases from and sells into its local market as defined by the EIA's Petroleum Administration for Defense Districts (PADD) region, as opposed to importing or exporting from other PADD region markets and incurring transportation costs in the process. These differences in energy commodity prices therefore have disparate impacts on the economic competitiveness of each location scenario.

The EIA publishes long-term price projections for each of the energy commodities listed in Table 4.2 as part of its AEO (19). Separate price projections are also made available by the EIA for each of the PADD regions. This analysis assumes that each cellulosic biofuel

facility will be operational for 20 years and it employs the EIA's projected prices for the years 2013-2032 to quantify the competitiveness of each location scenario. Furthermore, each location scenario uses the price projections for the relevant PADD scenario (e.g., Scenarios 4 and 5 employ the same projected energy commodity prices since Kansas and Iowa are both in the Midwest PADD region). The price of hydrogen is treated as a function of the natural gas price under the assumption that the former is produced via steam methane reforming of the latter (80).

4.3.7. Uncertainty analysis

This analysis employs Monte Carlo simulations to calculate both a 20-year NPV probability distribution and a 10-year (i.e., the assumed loan period) probability of default for each location scenario. The NPV probability distribution also makes it possible to derive the probability that each location scenario's 20-year NPV is greater than zero. This analysis employs the methodology developed by Brown and Wright (145) in previous work to conduct the uncertainty analysis, which consists of the following steps: (1) historical data on monthly energy commodity price changes is fit to probability distributions; (2) 20-year cash flows are deterministically simulated to calculate interest coverage ratios (ICR) for each location scenario; (3) costs of debt and equity are calculated for each location scenario as a function of their ICRs and industry betas, respectively; and (5) 20-year NPV distributions and 10-year probabilities of default are calculated for each location scenario. The methodology for calculating costs of debt and equity is covered in detail elsewhere (117,118,145) and therefore not repeated here.

The Monte Carlo simulation methodology employed in this analysis differs from previous work (104,145) in that it develops and employs energy commodity price

distributions that are unique to each location scenario. The probability distributions are developed based on the monthly price changes from January 2001 to December 2013 for natural gas, gasoline, diesel fuel, and LPG (the latter being a byproduct yielded by Scenario 8) in each location scenario's PADD region (146–148). EIA data on the historical gasoline and diesel fuel retail prices for each PADD region is adjusted to an ex-tax basis by excluding the federal and state fuel excise taxes for the respective location scenarios (92). The probability distribution for forest residue price changes for the scenarios employing that feedstock is derived from quarterly price data from January 1996 to December 2013 for Louisiana pine pulpwood (149). This probability distribution is applied to all of the location scenarios in which forestry residue is the assumed feedstock due to a lack of sufficient price data for other states. There is no historical data on corn stover prices from which to develop probability distributions due to the lack of an historical stover market. This analysis therefore uses historical monthly corn price changes (116) as a proxy from which to derive the probability distribution for stover due to stover being a byproduct of corn production. This analysis makes the assumption that the factors causing corn prices to increase (e.g., drought, late harvests, etc.) will also cause stover prices to increase, and vice versa. Finally, the Consumer Price Index is used to adjust all prices to a 2011 cost year basis prior to calculating their monthly changes (76). The probability distributions and monthly price change ranges within a 95% confidence interval for each energy commodity are presented in Table 4.3. While almost all of the energy commodity price changes are determined according to Anderson-Darling goodness-of-fit test results to best fit to logistic distributions (with the exception of natural gas under Scenario 2), the range of monthly price changes is found to differ according to energy commodity and location scenario.

Scenario	Commodity	Distribution	Monthly price
			change range (95%
			CI)
S1	Natural gas	Logistic	-18.1% - 18.5%
	Gasoline	Logistic	-18.3% - 18.4%
	Diesel fuel	Logistic	-10.9% - 11.1%
	Feedstock ^b	Logistic	-22.0% - 22.0%
S2	Natural gas	Student's t	-12.2% - 12.5%
	Gasoline	Logistic	-18.3% - 18.4%
	Diesel fuel	Logistic	-14.6% - 14.8%
	Feedstock ^b	Logistic	-22.0% - 22.0%
S3	Natural gas	Logistic	-10.8% - 11.2%
	Gasoline	Logistic	-14.6% - 14.7%
	Diesel fuel	Logistic	-14.6% - 14.8%
	Feedstock ^b	Logistic	-22.0% - 22.0%
S4	Natural gas	Logistic	-25.5% - 25.8%
	Gasoline	Logistic	-18.3% - 18.4%
	Diesel fuel	Logistic	-10.9% - 11.1%
	Feedstock ^a	Logistic	-11.0% - 11.0%
S5	Natural gas	Logistic	-25.5% - 25.9%
	Gasoline	Logistic	-18.3% - 18.4%
	Diesel fuel	Logistic	-10.9% - 11.1%
	Feedstock ^a	Logistic	-11.0% - 11.0%
S6	Natural gas	Logistic	-18.2% - 18.5%
	Gasoline	Logistic	-14.6% - 14.7%
	Diesel fuel	Logistic	-10.9% - 11.1%
	Feedstock ^a	Logistic	-11.0% - 11.0%
S7	Natural gas	Logistic	-18.1% - 18.5%
	Gasoline	Logistic	-18.3% - 18.4%
	Diesel fuel	Logistic	-10.9% - 11.1%
	Feedstock ^b	Logistic	-22.0% - 22.0%
S 8	Natural gas	Logistic	-25.4% - 25.9%
	Gasoline	Logistic	-14.6% - 14.7%
	Diesel fuel	Logistic	-10.9% - 11.1%
	LPG	Logistic	-20.7% - 20.9%
	Feedstock ^b	Logistic	-22.0% - 22.0%

Table 4.3. Probability distribution function for monthly change of each commodity price

^aDistribution based on historical corn prices due to lack of a developed stover market

^bDistribution based on historical Louisiana pine pulpwood prices due to lack of data from other states

Correlations between commodity prices are also identified and derived from the historical energy commodity prices for each PADD region included in this analysis. Strong correlations (>0.5) are identified via Spearman's correlation coefficient (150) for the monthly price changes of gasoline and diesel fuel, natural gas and LPG, and diesel fuel and corn for each PADD region. The prices of some energy commodities have been correlated historically due to their use of a common feedstock (e.g., gasoline and diesel fuel) or use as a similar product (e.g., natural gas and LPG). These correlations are included when performing the Monte Carlo simulations for the respective energy commodity price changes so as to prevent unrealistic price movements between energy commodity prices that have been historically correlated.

Future energy commodity prices are simulated via Monte Carlo simulation as a function of the EIA's price projections and the probability distributions based on historical monthly price changes. All of the simulated energy commodity prices with the exception of feedstock are assumed to follow a random walk with drift, where the drift is set according to the 20-year monthly price increase required for each price to reach the EIA's respective price projection for 2032. Feedstock price instead follows a random walk since the EIA does not provide 20-year price projections for stover and softwood. The monthly price changes are truncated to include only those falling within a 95% confidence interval (CI) so as to eliminate ahistorical extreme price movements. Crystal Ball® software is used to generate 10,000 random monthly price changes per energy commodity, with the price in one month equaling the previous month's price plus the random change, as presented by Esty (1999) (129):

$$P_t = P_{t-1}(1 + random change)$$

The amount of each random change is derived from the fitted probability distributions described above. When aggregated, the monthly price changes calculated by the Monte Carlo simulation yield 10,000 random price paths for each energy commodity and each location scenario over the 20-year facility life. These random price paths are incorporated into the DCFROR spreadsheet to calculate interest coverage ratios (ICR) for each location scenario via the methodology described in previous work (145). The ICRs are continuously adjusted until the cost of debt calculations converge, at which point the appropriate interest rate is incorporated into the location scenario (see Table 4.4) (117). The cost of equity is assumed to be 14.3% for all of the location scenarios based on the calculations by Brown and Wright (2014) (145).

Scenario	Mean interest coverage ratio	Cost of debt
S 1	0.12	18.7%
S2	-1.26	18.7%
S 3	-0.21	18.7%
S4	-0.98	18.7%
S5	-0.96	18.7%
S 6	-1.06	18.7%
S 7	-0.28	18.7%
S 8	1.74	9.0%

Table 4.4. Cost of debt results for each location scenario

10-year probabilities of default are calculated stochastically as the probability that the mean minimum debt coverage service ratio (DCSR) of a location scenario is less than 1.0. The methodology for calculating DCSR is covered elsewhere (129,145) and not repeated here. This analysis increases the capital cost of each location scenario in the form of cash working capital so that the DSCR in the first six months of operations is equal to the greater of 1.0 or the mean DCSR over the first 10 years of operations. This adjustment is made to prevent the minimum DSCR from falling below 1.0 (and generating a default) during three

years of construction and the six-month start-up process, the latter period during which revenues are assumed to be only 50% of normal. The percentage of working capital relative to each location scenario's capital cost is therefore a function of both its cost of debt and its total capital cost, although this analysis assumes that it does not fall below 15%, which is the value commonly assumed in the TEAs used to develop the eight location scenarios.

4.4. Results

20-year NPVs and 10-year probabilities of default are stochastically calculated for six cellulosic biofuel pathways under a total of eight location scenarios. Seven of the eight scenarios have very low probabilities of achieving a 20-year NPV that is greater than zero while only Scenario 8 has a greater than 50% probability of achieving the threshold (see Table 4.5). The other seven scenarios all have probabilities of below 5% of achieving the threshold. One of the main drivers of negative NPVs is the high costs of debt incurred by most of the location scenarios (see Table 4.4). These costs necessitate high amounts (both in absolute terms and relative to capital costs) of working capital to provide Scenarios 1-7 with sufficient DSCRs during their construction and start-up periods. The working capital contributes to each scenario's total liabilities, however, causing interest payments to rise and NPV to fall. With the exception of Scenario 8, none of the scenarios achieve positive 20-year NPVs in the absence of sustained very low input costs and very high output market prices.

Scenario	Mean NPV (\$MM)	Prob. NPV>0	Default prob.
S 1	-588.0	4.6%	98.9%
S2	-1,401.1	0.1%	100.0%
S 3	-1,738.3	4.6%	98.7%
S4	-1,025.3	1.8%	100.0%

Table 4.5. Location scenario mean 20-year NPVs and 10-year probabilities of default.

Table 4.5 continued					
Scenario	Mean NPV (\$MM)	Prob. NPV>0	Default prob.		
S5	-1,093.0	1.1%	100.0%		
S6	-940.1	0.7%	100.0%		
S7	-1,022.9	0.2%	100.0%		
S8	33.0	54.2%	61.1%		

Scenarios 1-7 all have very high (>98%) 10-year default probabilities (see Table 4.5). The majority of the scenarios have a 100% probability of defaulting during the first decade of operations, while even Scenario 8 has a greater than even probability of doing so. As with the NPV calculations, the 10-year default probabilities are driven by the high costs of debt incurred by the majority of the location scenarios relative to their probable revenues. A negative feedback loop is witnessed in Scenarios 1-7 in which high capital costs incur high costs of debt, necessitating higher capital costs to keep the DSCR above 1.0 during the construction and start-up periods. While this increase to absolute debt costs ultimately causes the default probability to approach 100%, it is necessary if default is not to occur prior to the commencement of full operations.

The results of this analysis indicate that the first several cellulosic biofuel pathways to be commercialized within the U.S. will not be competitive with fossil fuels on an unsubsidized basis. Indeed, it is more likely than not that the first facilities to employ these pathways will default prior to achieving a decade of continuous operations unless supported by the government. Faced with such low probabilities of success and substantial regulatory uncertainty regarding the future shape and existence of the RFS2, it should come as little surprise that U.S. cellulosic biofuel capacity investment to date has fallen well short of its mandated volume. Furthermore, given the historically-low interest rates and relatively high petroleum prices that have prevailed in the U.S. over the last five years, it is unlikely that a
sudden change in macroeconomic conditions will improve the future competitiveness of the pathways included in this analysis. Future research should focus on reducing pathway costs of capital by improving product yields and, perhaps more importantly, diversifying product portfolios so as to reduce debt costs by decreasing the sensitivity of pathway cash flows to a single energy commodity price.

4.5. Conclusion

This paper quantifies the 20-year net present value (NPV) and 10-year probability of default for six cellulosic biofuel pathways under eight location scenarios. Each location scenario is designed to resemble one of the commercial-scale cellulosic biofuel projects planned within the U.S. at the end of 2012 and accounts for location-specific factors such as capital costs, feedstock type, feedstock costs, energy commodity prices, and state corporate income tax rates. Furthermore, scenario-specific costs of debt and costs of equity are employed. This paper stochastically simulates 20-year cash flows via Crystal Ball software and a discounted cash flow rate of return spreadsheet to calculate the NPV and probability of default for each location scenario.

Negative mean 20-year NPVs are calculated for seven of the eight location scenarios. A positive 20-year mean NPV of \$33 million is calculated for the eighth scenario, which is also the only scenario to have a probability of a positive NPV that exceeds 50%. Very high (>98%) 10-year probabilities of default are also calculated for seven of the eight scenarios. The eighth scenario also has a probability of default that is greater than 50%, although at 61.1% it is the lowest of the eight location scenarios.

CHAPTER 5. CONCLUSIONS AND RECOMMENDATIONS FOR FUTURE WORK

5.1. Conclusions

The revised Renewable Fuel Standard (RFS2) requires the annual blending of 16 billion gallons of cellulosic biofuel with conventional fuels by 2022. It employs two mechanisms to ensure that this goal is met: a binding blending mandate and a blending subsidy in the form of Renewable Identification Numbers (RIN). Despite these mechanisms, a combination of technical, financial, and economic hurdles have prevented the cellulosic biofuels industry from producing more than a tiny fraction of the volumes required to be blended under the mandate to date. This underproduction has occurred despite the publication of several studies in the refereed literature indicating that some cellulosic biofuel pathways will be competitive with petroleum on an unsubsidized basis after commercialization. Furthermore, a lack of investment in cellulosic biofuels capacity has prevented researchers from accurately assessing the likely future impacts, both economic and otherwise, of the cellulosic biofuels mandate since the identities of the feedstocks, pathways, and locations most likely to be involved are not yet known. This dissertation develops a framework for identifying and analyzing the likely composition of the cellulosic biofuels industry in the future. Finally, it also uses the framework to provide a possible explanation for why investment in cellulosic biofuels capacity has been low to date despite the presence of the RFS2 and techno-economic analyses (TEA) suggesting that some pathways will be competitive on an unsubsidized basis following commercialization.

The results of this dissertation indicate that few cellulosic biofuel pathways will be economically competitive with petroleum on an unsubsidized basis. Of five cellulosic biofuel pathways considered under 20-year price forecasts with volatility, only two (fast pyrolysis

and hydroprocessing; gasification and methanol-to-gasoline synthesis) achieve positive mean 20-year net present value (NPV) probabilities. Furthermore, recent exploitation of U.S. shale gas reserves and the subsequent fall in U.S. natural gas prices have negatively impacted the economic competitiveness of all but two of the cellulosic biofuel pathways considered; only two of the five pathways (fast pyrolysis and hydroprocessing; gasification and acetic acid synthesis) achieve substantially higher 20-year NPVs under a post-shale gas economic scenario relative to a pre-shale gas scenario.

The economic competitiveness of cellulosic biofuel pathways with petroleum is reduced further when considered under price uncertainty in combination with realistic financial assumptions. While many cellulosic biofuel TEAs in the refereed literature assume the same cost of capital (10% cost of equity and no debt financing), this dissertation employs financial analysis tools to calculate pathway-specific costs of capital for five cellulosic biofuel pathway scenarios. The analysis finds that most of the pathways incur a 14.3% cost of equity and an 18.7% cost of debt based on their cash flow volatility, both of which are substantially higher than the costs commonly assumed in the literature. Employment of these costs of capital in a comparative TEA greatly reduces the mean 20-year NPVs for each pathway while increasing their 10-year probabilities of default. Default probabilities in excess of 80% are calculated for all five scenarios analyzed on U.S. Gulf Coast and unsubsidized bases under national energy commodity prices.

Finally, this dissertation finds that the use of a U.S. Gulf Coast basis and national energy commodity prices may result in the overestimation of the economic competitiveness of cellulosic biofuel pathways. The economic competitiveness of six cellulosic biofuel pathways currently being commercialized in eight different U.S. states is quantified under

price uncertainty, utilization of pathway-specific costs of capital, and region-specific economic factors. 10-year probabilities of default in excess of 60% are calculated for all eight location scenarios considered, with default probabilities in excess of 98% calculated for seven of the eight. Furthermore, negative mean 20-year NPVs are calculated for seven of the eight location scenarios, with only one scenario (gasification and methanol-to-gasoline synthesis in Louisiana) generating a positive mean 20-year NPV at \$33 million.

5.2. Recommendations for Future Work

This dissertation develops a framework of methodologies for quantifying the economic competitiveness of multiple cellulosic biofuel pathways under price volatility, price uncertainty, scenario-specific costs of capital, and regional factors. Its presentation of both the methodologies employed as well as the results generated permits its continued application as the cellulosic biofuel industry grows and evolves, both in the U.S. and worldwide. The majority of the cellulosic biofuel TEAs currently found in the literature utilizes generic financial assumptions across different pathways and presents the economic competitiveness results as a single number, the implication being that this result is known with complete certainty. Furthermore, those TEAs that do include uncertainty analysis via Monte Carlo simulation frequently employ default probability distributions for the economic variables rather than distributions developed from historical prices. Finally, most TEAs calculate facility capital costs on a U.S. Gulf Coast basis by default and utilize national energy commodity prices to calculate cash flows. The methodologies presented in this dissertation increase the accuracy of cellulosic biofuel TEA results by employing more realistic cost assumptions and factors.

An important aspect of cellulosic biofuel TEAs that is not considered by this dissertation is processing uncertainty. Feedstock yields, process yields, and product yields are all important drivers of a pathway's economic competitiveness. All three are also uncertain, especially for pathways that have yet to begin or are undergoing the early stages of commercialization. In order for the yield uncertainty analysis to be accurate, however, the probability distributions should be based on experimental values. At present experimental yields are commonly reported in triplicate form, which is an insufficient number of data points for the development of probability distributions. The generation of a larger number of yield data points for an individual pathway would permit the inclusion of feedstock and pathway yields in the uncertainty analysis methodology presented in this dissertation, further increasing the accuracy and value of the economic competitiveness assessments.

The regional framework presented in Chapter 4 could be further expanded by consideration of region-specific differences in consumer acceptance to biofuels in general and different biofuel types in particular. For example, Midwestern consumers are more willing to purchase fuels containing higher blends of ethanol than are their counterparts near the East and West Coasts. Similarly, consumers might be willing to pay a premium for cellulosic hydrocarbon-based fuels that they are not willing to pay for cellulosic ethanol, and this premium could vary by region. Consumer acceptance of ethanol will be an important factor in the future composition of the cellulosic biofuel industry due to the existence of the ethanol blend wall. Furthermore, whether consumers require a discount or are willing to pay a premium for one fuel type relative to another will affect the economic competitiveness of cellulosic biofuel pathways. Further research into the consumer acceptance of cellulosic

biofuels would further increase the accuracy and value of results generated via the methodology presented in this dissertation.

The methodology presented in this dissertation can also be employed to conduct further analysis into the likely economic, environmental, ecological, and socioeconomic impacts of the RFS2's cellulosic biofuels mandate. At present the likely economic costs of the cellulosic biofuels mandate are unknown due to a lack of commercialization to date. Furthermore, TEAs of cellulosic biofuel pathways in the literature overestimate certainty and frequently employ arbitrary assumptions, preventing comparisons with other pathway TEAs. As such, the identities, costs, and economic competitiveness of the pathways most likely to be employed remain unknown, preventing efforts to calculate the subsidies under the RFS2 that will be necessary for the industry to succeed. Furthermore, in the initial years of commercial-scale production the RFS2 is intended to incentivize all of the pathways generating cellulosic biofuels by the amount necessary to make the least-competitive pathway competitive with petroleum. A calculated RIN subsidy value can therefore be utilized within the framework to calculate the economic competitiveness of the pathways on a subsidized basis.

Similarly, the lack of information regarding which of the available pathways are most likely to be competitive and therefore productive in the near-term also prevents accurate assessments of the mandate's environmental, ecological, and socioeconomic impacts. As discussed in Chapter 1, these impacts vary by pathway: the impacts of a cellulosic biofuel industry that is dominated by the conversion of stover to cellulosic ethanol via enzymatic hydrolysis will be different from an industry comprised of multiple pathways utilizing mixed feedstocks to produce both ethanol and hydrocarbon-based biofuels. By providing a

methodology for the identification of the feedstock, pathway, and location combinations that are most likely to be economically competitive in the U.S., this dissertation enables more accurate analyses of the mandate's potential impacts.

REFERENCES

- 1. Anon. Historic U.S. Fuel Ethanol Production [Internet]. Renewable Fuels Association Statistics. Renewable Fuels Association; 2014 [cited 2014 Mar 20]. Available from: http://www.ethanolrfa.org/pages/statistics
- 2. Schnepf R, Yacobucci BD. Renewable Fuel Standard (RFS): Overview and Issues. Washington, DC: Congressional Research Service; 2013.
- 3. Runge CF, Senauer B. How Biofuels Could Starve the Poor. Foreign Aff. 2007;86(3):41–53.
- 4. Searchinger T, Heimlich R, Houghton R a, Dong F, Elobeid A, Fabiosa J, et al. Use of U.S. croplands for biofuels increases greenhouse gases through emissions from land-use change. Science (80-). 2008 Feb 29;319(5867):1238–40.
- 5. Anon. 2013 EMTS Data [Internet]. EPA; 2013 [cited 2014 Apr 2]. Available from: http://www.epa.gov/otaq/fuels/rfsdata/2013emts.htm
- 6. Solomon BD, Barnes JR, Halvorsen KE. Grain and cellulosic ethanol: History, economics, and energy policy. Biomass and Bioenergy. 2007;31(6):416–25.
- Songstad D, Lakshmanan P, Chen J, Gibbons W, Hughes S, Nelson R. Historical Perspective of Biofuels: Learning from the Past to Rediscover the Future. In: Tomes D, Lakshmanan P, Songstad D, editors. Biofuels. New York, NY: Springer New York; 2011. p. 1–7.
- 8. Morgan D. Senate Panel Votes to Boost Ethanol Mandate. The Washington Post. Washington D.C.; 2005 May 26;
- 9. Babcock BA. Mandates, Tax Credits, and Tariffs: Does the U.S. Biofuels Industry Need them All? Ames, IA: Center for Agricultural and Rural Development; 2010.
- 10. Brown RC, Brown TR. Biorenewable Resources: Engineering New Products from Agriculture. 2nd ed. Oxford: Wiley Blackwell; 2014.
- 11. Anon. 2011 EMTS Data [Internet]. EPA; 2011 [cited 2014 Apr 1]. Available from: http://www.epa.gov/otaq/fuels/rfsdata/2011emts.htm
- 12. Anon. 2012 EMTS Data [Internet]. EPA; 2012 [cited 2014 Apr 1]. Available from: http://www.epa.gov/otaq/fuels/rfsdata/2012emts.htm
- Service RF. Court Strikes Down EPA Renewable Fuel Rule | Science/AAAS | News [Internet]. Science. 2013 [cited 2013 Aug 18]. Available from: http://news.sciencemag.org/policy/2013/01/court-strikes-down-epa-renewable-fuelrule

- 14. U.S. Court of Appeals for the District of Columbia Circuit. American Petroleum Institute v. Environmental Protection Agency. Washington D.C.; 2013.
- 15. Wald M. E.P.A. Approves Use of More Ethanol in Gasoline. The New York Times. Late. Washington, DC; 2011;B7.
- 16. Wald M. In Kansas, Stronger Mix Of Ethanol. The New York Times. Late. Lawrence, Kan.; 2012;B1.
- 17. Westbrook J, Barter GE, Manley DK, West TH. A parametric analysis of future ethanol use in the light-duty transportation sector: Can the US meet its Renewable Fuel Standard goals without an enforcement mechanism? Energy Policy. 2014 Feb;65:419–31.
- 18. Anon. Annual Energy Outlook 2008. Washington D.C.; 2008.
- 19. Anon. Annual Energy Outlook 2013. Washington, DC; 2012.
- 20. McPhail L, Westcott P, Lutman H. The Renewable Identification Number System and U.S. Biofuel Mandates. Washington, DC: Economic Research Service; 2011.
- 21. Morgenson G, Gebeloff R. Wall St. Exploits Ethanol Credits, and Prices Spike. The New York Times. New York City; 2013;A1.
- 22. Anon. RIN Index [Internet]. EcoEngineers. Des Moines: EcoEngineers; 2014 [cited 2014 Mar 21]. Available from: http://rinindex.ecoengineers.us/
- Paulson N. EPA Announces 2013 RFS2 Final Rulemaking [Internet]. Urbana-Champaign; 2013 [cited 2014 Apr 1]. Available from: http://farmdocdaily.illinois.edu/2013/08/epa-announces-2013-rfs2-finalrulemaking.html
- 24. Anon. EPA Proposes 2014 Renewable Fuel Standards, 2015 Biomass-Based Diesel Volume [Internet]. EPA; 2013. Available from: http://www.epa.gov/OTAQ/fuels/renewablefuels/documents/420f13048.pdf
- 25. Babcock B a., Marette S, Tréguer D. Opportunity for profitable investments in cellulosic biofuels. Energy Policy. 2011 Feb;39(2):714–9.
- 26. DOE. U.S. Billion-Ton Update: Biomass Supply for a Bioenergy and Bioproducts Industry. Oak Ridge, TN: Oak Ridge National Laboratory; 2011.
- 27. Brown TR, Brown RC. A review of cellulosic biofuel commercial-scale projects in the United States. Biofuels, Bioprod Biorefining. John Wiley & Sons, Ltd; 2013;7(3):235–45.

- Wright MM, Daugaard DE, Satrio J a., Brown RC. Techno-economic analysis of biomass fast pyrolysis to transportation fuels. Fuel. Elsevier Ltd; 2010 Nov;89:S2– S10.
- 29. Swanson RM, Platon A, Satrio JA, Brown RC. Techno-economic analysis of biomass-to-liquids production based on gasification. Fuel. 2010;89(Supplement 1):S11–S19.
- Kazi FK, Fortman J a., Anex RP, Hsu DD, Aden A, Dutta A, et al. Techno-economic comparison of process technologies for biochemical ethanol production from corn stover. Fuel. Elsevier Ltd; 2010 Nov;89:S20–S28.
- 31. Anex RP, Aden A, Kazi FK, Fortman J, Swanson RM, Wright MM, et al. Technoeconomic comparison of biomass-to-transportation fuels via pyrolysis, gasification, and biochemical pathways. Fuel. Elsevier; 2010;89(Supplement 1):S29–S35.
- 32. Brown TR, Thilakaratne R, Brown RC, Hu G. Regional differences in the economic feasibility of advanced biorefineries: Fast pyrolysis and hydroprocessing. Energy Policy. 2013;57:234–43.
- Spatari S, Bagley DM, MacLean HL. Life cycle evaluation of emerging lignocellulosic ethanol conversion technologies. Bioresour Technol. Elsevier Ltd; 2010 Jan;101(2):654–67.
- 34. Zhang Y, Hu G, Brown RC. Life cycle assessment of the production of hydrogen and transportation fuels from corn stover via fast pyrolysis. Environ Res Lett. 2013;8(2):25001.
- 35. Hsu DD. Life cycle assessment of gasoline and diesel produced via fast pyrolysis and hydroprocessing. Biomass and Bioenergy. 2012;45(0):41–7.
- 36. Han J, Elgowainy A, Dunn JB, Wang MQ. Life cycle analysis of fuel production from fast pyrolysis of biomass. Bioresour Technol. 2013 Apr;133:421–8.
- 37. Zaman AU. Comparative study of municipal solid waste treatment technologies using life cycle assessment method. Int J Environ Sci Tech. 2010;7(2):225–34.
- Van Vliet OPR, Faaij APC, Turkenburg WC. Fischer–Tropsch diesel production in a well-to-wheel perspective: A carbon, energy flow and cost analysis. Energy Convers Manag. Elsevier Ltd; 2009 Apr;50(4):855–76.
- Hsu DD, Inman D, Heath GA, Wolfrum EJ, Mann MK, Aden A. Life Cycle Environmental Impacts of Selected U.S. Ethanol Production and Use Pathways in 2022. Environ Sci Technol. American Chemical Society; 2010;44(13):5289–97.

- 40. Tao L, Tan ECD, McCormick R, Zhang M, Aden A, He X, et al. Techno-economic analysis and life-cycle assessment of cellulosic isobutanol and comparison with cellulosic ethanol and n-butanol. Biofuels, Bioprod Biorefining. 2013 Sep 2;n/a–n/a.
- 41. Anon. Renewable Fuel Standard Program (RFS2) Regulatory Impact Analysis. Washington D.C.; 2010.
- 42. Anon. Regulation of Fuels and Fuel Additives: Identification of Additional Qualifying Renewable Fuel Pathways Under the Renewable Fuel Standard Program. Fed Regist. 2013;78(43):14190–217.
- 43. McKechnie J, Zhang Y, Ogino A, Saville B, Sleep S, Turner M, et al. Impacts of colocation, co-production, and process energy source on life cycle energy use and greenhouse gas emissions of lignocellulosic ethanol. Biofuels, Bioprod Biorefining. John Wiley & Sons, Ltd.; 2011;5(3):279–92.
- Gelfand I, Sahajpal R, Zhang X, Izaurralde RC, Gross KL, Robertson GP. Sustainable bioenergy production from marginal lands in the US Midwest. Nature. Nature Publishing Group, a division of Macmillan Publishers Limited. All Rights Reserved.; 2013 Jan 24;493(7433):514–7.
- 45. Baker JS, McCarl B, Murray BC, Rose SK, Alig R, Adams D, et al. Net Farm Income and Land Use under a U.S. Greenhouse Gas Cap and Trade. Policy Issues. 2010;P17:1–5.
- 46. Elobeid A, Carriquiry MA, Dumortier J, Rosas F, Mulik K, Fabiosa J, et al. Biofuel Expansion, Fertilizer Use, and GHG Emissions: Unintended Consequences of Mitigation Policies. Econ Res Int. 2013;2013:1–12.
- 47. Fargione JE, Cooper TR, Flaspohler DJ, Hill J, Lehman C, Tilman D, et al. Bioenergy and Wildlife: Threats and Opportunities for Grassland Conservation. Bioscience. Oxford University Press; 2009 Oct 1;59(9):767–77.
- 48. Wilhelm WW, Johnson JMF, Karlen DL, Lightle DT. Corn Stover to Sustain Soil Organic Carbon Further Constrains Biomass Supply. Agron J. American Society of Agronomy; 2007 Nov 1;99(6):1665.
- 49. Johnson J, Coleman M, Gesch R, Jaradat A, Mitchell R, Reicosky D, et al. Biomassbioenergy crops in the United States: A changing paradigm. Am J Plant Sci Biotechnol. 2007;1(1):1–28.
- 50. Karlen DL, Birell SJ, Hess JR. A five-year assessment of corn stover harvest in central Iowa, USA. Soil Tillage Res. 2011 Oct;115-116:47–55.
- 51. Simpson TW, Sharpley AN, Howarth RW, Paerl HW, Mankin KR. The new gold rush: fueling ethanol production while protecting water quality. J Environ Qual.

American Society of Agronomy, Crop Science Society of America, Soil Science Society; 2008 Jan 1;37(2):318–24.

- 52. Rossi AM, Hinrichs CC. Hope and skepticism: Farmer and local community views on the socio-economic benefits of agricultural bioenergy. Biomass and Bioenergy. 2011 Apr;35(4):1418–28.
- 53. Hall J, Matos S, Severino L, Beltrão N. Brazilian biofuels and social exclusion: established and concentrated ethanol versus emerging and dispersed biodiesel. J Clean Prod. 2009 Nov;17:S77–S85.
- 54. Wegener DT, Kelly JR. Social Psychological Dimensions of Bioenergy Development and Public Acceptance. BioEnergy Res. 2008 Jul 16;1(2):107–17.
- 55. Anon. U.S. Natural Gas Marketed Production [Internet]. Energy Information Administration. Energy Information Administration; 2013 [cited 2014 Apr 2]. Available from: http://www.eia.gov/dnav/ng/hist/n9050us2a.htm
- 56. EIA. U.S. Natural Gas Wellhead Price [Internet]. Energy Information Administration. Energy Information Agency; 2013 [cited 2014 Apr 2]. Available from: http://www.eia.gov/dnav/ng/hist/n9190us3m.htm
- 57. EIA. Cushing, OK Crude Oil Future Contract 1 [Internet]. Energy Information Administration. Energy Information Administration; 2013 [cited 2014 Mar 30]. Available from: http://www.eia.gov/dnav/pet/hist/LeafHandler.ashx?n=PET&s=RCLC1&f=M
- 58. Anon. Dry Natural Gas Proved Reserves [Internet]. Energy Information Administration. Energy Information Administration; 2012 [cited 2014 Mar 20]. Available from: http://www.eia.gov/dnav/ng/ng_enr_dry_a_EPG0_r11_bcf_a.htm
- 59. Anon. Shale Gas [Internet]. Energy Information Administration. Energy Information Administration; 2012 [cited 2014 Apr 2]. Available from: http://www.eia.gov/dnav/ng/ng_enr_shalegas_a_EPG0_R5301_Bcf_a.htm
- 60. Bevill K. Bill proposes fossil fuel-derived ethanol should qualify for RFS. Ethanol Producer Magazine. 2012.
- 61. Herndon A, Swint B. Shale Glut Becomes \$2 Diesel Using Gas-to-Liquids Plants. Bloomberg. Bloomberg; 2012.
- 62. Proctor C. Rentech shutting Commerce City plant. Denver Business Journal. Denver; 2013.
- 63. Forman GS, Hahn TE, Jensen SD. Greenhouse Gas Emission Evaluation of the GTL Pathway. Environ Sci Technol. American Chemical Society; 2011;45(20):9084–92.

- 64. Jaramillo P, Griffin WM, Matthews HS. Comparative Analysis of the Production Costs and Life-Cycle GHG Emissions of FT Liquid Fuels from Coal and Natural Gas. Environ Sci Technol. American Chemical Society; 2008;42(20):7559–65.
- 65. Van Vliet OPR, Faaij APC, Turkenburg WC. Fischer–Tropsch diesel production in a well-to-wheel perspective: A carbon, energy flow and cost analysis. Energy Convers Manag. 2009 Apr;50(4):855–76.
- 66. Regalbuto JR. The sea change in US biofuels' funding: from cellulosic ethanol to green gasoline. Biofuels, Bioprod Biorefining. John Wiley & Sons, Ltd.; 2011;5(5):495–504.
- Lemus RG, Martínez Duart JM. Updated hydrogen production costs and parities for conventional and renewable technologies. Int J Hydrogen Energy. 2010;35(9):3929– 36.
- 68. Balat M, Balat M. Political, economic and environmental impacts of biomass-based hydrogen. Int J Hydrogen Energy. 2009;34(9):3589–603.
- 69. Brown TR, Thilakaratne R, Brown RC, Hu G. Techno-economic analysis of biomass to transportation fuels and electricity via fast pyrolysis and hydroprocessing. Fuel. Elsevier Ltd; 2013 Apr;106:463–9.
- 70. Jones SB, Valkenburg C, Walton CW, Elliot DC, Holladay JE, Stevens DJ, et al. Production of Gasoline and Diesel from Biomass via Fast Pyrolysis, Hydrotreating and Hydrocracking: A Design Case. Richland: Pacific Northwest National Laboratory; 2009.
- 71. Zhu Y, Jones SB. Techno-economic analysis for the thermochemical conversion of lignocellulosic biomass to ethanol via acetic acid synthesis. Richland: Pacific Northwest National Laboratory; 2009.
- 72. Kazi FK, Fortman JA, Anex RP, Hsu DD, Aden A, Dutta A, et al. Techno-economic comparison of process technologies for biochemical ethanol production from corn stover. Fuel. 2010;89(Supplement 1):S20–S28.
- 73. Phillips SD, Tarud JK, Biddy MJ, Dutta A. Gasoline from wood via integrated gasification, synthesis, and methanol-to-gasoline technologies. Golden: National Renewable Energy Laboratory; 2011.
- 74. Phillips SD, Tarud JK, Biddy MJ, Dutta A. Gasoline from Woody Biomass via Thermochemical Gasification, Methanol Synthesis, and Methanol-to-Gasoline Technologies: A Technoeconomic Analysis. Ind Eng Chem Res. American Chemical Society; 2011;50(20):11734–45.

- 75. Wright M, Brown RC. Establishing the optimal sizes of different kinds of biorefineries. Biofuels, Bioprod Biorefining. John Wiley & Sons, Ltd.; 2007;1(3):191–200.
- 76. Anon. Consumer Price Index [Internet]. Bureau of Labor Statistics; 2014 [cited 2014 Apr 1]. Available from: ftp://ftp.bls.gov/pub/special.requests/cpi/cpiai.txt
- 77. Kazi FK, Patel AD, Serrano-Ruiz JC, Dumesic JA, Anex RP. Techno-economic analysis of dimethylfuran (DMF) and hydroxymethylfurfural (HMF) production from pure fructose in catalytic processes. Chem Eng J. 2011;169(1–3):329–38.
- 78. Dutta A, Bain RL, Biddy MJ. Techno-economics of the production of mixed alcohols from lignocellulosic biomass via high-temperature gasification. Environ Prog Sustain Energy. John Wiley & Sons, Inc.; 2010;29(2):163–74.
- 79. Anon. Annual Energy Outlook 2010 with Projections to 2035. Washington DC; 2010.
- Steward D, Ramsden T, Zuboy J. H2A Central Hydrogen Production Model, Version 3. U.S. Department of Energy; 2012.
- 81. Chou J-S, Yang IT, Chong WK. Probabilistic simulation for developing likelihood distribution of engineering project cost. Autom Constr. 2009;18(5):570–7.
- 82. Spinney PJ, Watkins GC. Monte Carlo simulation techniques and electric utility resource decisions. Energy Policy. 1996;24(2):155–63.
- Silva C, Vale S, Pinto FC, Müller CS, Moura A. The economic feasibility of precision agriculture in Mato Grosso do Sul State, Brazil: a case study. Precis Agric. Springer US; 2007;8(6):255–65.
- 84. Zhang Y, Brown TR, Hu G, Brown RC. Techno-economic analysis of two bio-oil upgrading pathways. Chem Eng J. 2013;225:895–904.
- Zhang Y, Brown TR, Hu G, Brown RC. Techno-economic analysis of monosaccharide production via fast pyrolysis of lignocellulose. Bioresour Technol. Elsevier Ltd; 2013;127:358–65.
- 86. Zhang Y, Brown TR, Hu G, Brown RC. Comparative techno-economic analysis of biohydrogen production via bio-oil gasification and bio-oil reforming. Biomass and Bioenergy. 2013;(0).
- Shlyakhter AI, Kammen DM, Broido CL, Wilson R. Quantifying the credibility of energy projections from trends in past data: The US energy sector. Energy Policy. 1994;22(2):119–30.

- 88. Mantripragada HC, Rubin ES. Techno-economic evaluation of coal-to-liquids (CTL) plants with carbon capture and sequestration. Energy Policy. 2011;39(5):2808–16.
- Mantripragada HC, Rubin ES. Performance, cost and emissions of coal-to-liquids (CTLs) plants using low-quality coals under carbon constraints. Fuel. 2013;103(0):805–13.
- Babonneau F, Haurie A, Loulou R, Vielle M. Combining Stochastic Optimization and Monte Carlo Simulation to Deal with Uncertainties in Climate Policy Assessment. Environ Model Assess. Springer Netherlands; 2012;17(1-2):51–76.
- 91. Stephens MA. EDF statistics for goodness of fit and some comparisons. J Am Stat Assoc. 1974;69(347):730–7.
- 92. API. State Motor Fuel Excise Tax Report. Washington DC: American Petroleum Institute; 2014.
- 93. Wright MM, Daugaard DE, Satrio JA, Brown RC. Techno-economic analysis of biomass fast pyrolysis to transportation fuels. Fuel. 2010;89(Supplement 1):S2–S10.
- 94. Wright MM, Satrio JA, Brown RC. Techno-economic analysis of biomass fast pyroysis to transportation fuels. Golden, CO: National Renewable Energy Laboratory; 2010.
- 95. EPA. Regulation of Fuels and Fuel Additive Changes to Renewable Fuel Standard Program; Final Rule. Fed Regist. 2010;75(58):14790.
- 96. Zhang Y-HP, Lynd LR. Toward an aggregated understanding of enzymatic hydrolysis of cellulose: Noncomplexed cellulase systems. Biotechnol Bioeng. Wiley Subscription Services, Inc., A Wiley Company; 2004;88(7):797–824.
- 97. Elgin B, Waldman P. Chevron Defies California On Carbon Emissions Bloomberg [Internet]. Bloomberg. 2013 [cited 2013 Jul 24]. Available from: http://www.bloomberg.com/news/2013-04-18/chevron-defies-california-on-carbonemissions.html
- 98. Downing L, Gismatullin E. Biofuel Investments at Seven-Year Low as BP Blames Cost - Bloomberg [Internet]. Bloomberg. 2013 [cited 2013 Jul 23]. Available from: http://www.bloomberg.com/news/2013-07-07/biofuel-investments-at-seven-year-lowas-bp-blames-cost.html
- CEEIIBP, NRC. Renewable Fuel Standard: Potential Economic and Environmental Effects of U.S. Biofuel Policy. Washington, D.C.: The National Academies Press; 2011.

- 100. Tan ECD, Marker TL, Roberts MJ. Direct production of gasoline and diesel fuels from biomass via integrated hydropyrolysis and hydroconversion process—A technoeconomic analysis. Environ Prog Sustain Energy. 2013;
- Phillips SD. Technoeconomic Analysis of a Lignocellulosic Biomass Indirect Gasification Process To Make Ethanol via Mixed Alcohols Synthesis. Ind Eng Chem Res. American Chemical Society; 2007;46(26):8887–97.
- 102. Brown TR, Wright MM, Brown RC. Estimating profitability of two biochar production scenarios: slow pyrolysis vs fast pyrolysis. Biofuels, Bioprod Biorefining. John Wiley & Sons, Ltd.; 2011;5(1):54–68.
- 103. Brown TR, Zhang Y, Hu G, Brown RC. Techno-economic analysis of biobased chemicals production via integrated catalytic processing. Biofuels, Bioprod Biorefining. 2012;6(1):73–87.
- 104. Brown TR, Wright MM. Techno-economic impacts of shale gas on cellulosic biofuel pathways. Fuel. Elsevier Ltd; 2014 Jan;117(April 2012):989–95.
- 105. Shah A, Darr M. Techno-economic analysis and life cycle assessment of the corn stover biomass feedstock supply chain system for a Midwest-based first-generation cellulosic biorefinery. Iowa State University; 2013. p. 34–136.
- Petrolia DR. The economics of harvesting and transporting corn stover for conversion to fuel ethanol: A case study for Minnesota. Biomass and Bioenergy. 2008 Jul;32(7):603–12.
- 107. Kadam KL, McMillan JD. Availability of corn stover as a sustainable feedstock for bioethanol production. Bioresour Technol. 2003;88(1):17–25.
- 108. Leboreiro J, Hilaly AK. Biomass transportation model and optimum plant size for the production of ethanol. Bioresour Technol. 2011;102(3):2712–23.
- 109. O'Neill BC, Desai M. Accuracy of past projections of US energy consumption. Energy Policy. 2005;33(8):979–93.
- 110. Craig PP, Gadgil A, Koomey JG. WHAT CAN HISTORY TEACH US? A Retrospective Examination of Long-Term Energy Forecasts for the United States*. Annu Rev Energy Environ. Annual Reviews 4139 El Camino Way, P.O. Box 10139, Palo Alto, CA 94303-0139, USA; 2002 Nov 28;27(1):83–118.
- 111. Anon. Henry Hub Gulf Coast Natural Gas Spot Price [Internet]. Energy Information Administration. EIA; 2013 [cited 2014 Mar 20]. Available from: http://www.eia.gov/dnav/ng/hist/rngwhhdm.htm

- 112. Balat M. Possible Methods for Hydrogen Production. Energy Sources, Part A Recover Util Environ Eff. Taylor & Francis; 2008 Dec 2;31(1):39–50.
- Anon. Average retail price of electricity to ultimate customers [Internet]. Energy Information Administration. Energy Information Administration; 2013 [cited 2014 Mar 19]. Available from: http://www.eia.gov/electricity/monthly/epm table grapher.cfm?t=epmt 5 3
- 114. Anon. Los Angeles, CA Ultra-Low Sulfur CARB Diesel Spot Price [Internet]. Energy Information Administration. EIA; 2013 [cited 2013 Mar 20]. Available from: http://www.eia.gov/dnav/pet/hist/LeafHandler.ashx?n=PET&s=EER_EPD2DC_PF4_ Y05LA_DPG&f=M
- 115. Anon. U.S. Gulf Coast Conventional Gasoline Regular Spot Price FOB [Internet]. EIA; 2013 [cited 2014 Jan 2]. Available from: http://www.eia.gov/dnav/pet/hist/LeafHandler.ashx?n=PET&s=EER_EPMRU_PF4_ RGC_DPG&f=M
- 116. Anon. US Average Farm Price Received Database [Internet]. FarmDoc. Champaign-Urbana: University of Illinois; 2013 [cited 2014 Mar 1]. Available from: http://farmdoc.illinois.edu/manage/uspricehistory/us_price_history.html
- 117. Damodaran A. Estimating Risk Paramters and Costs of Financing. Investment Valuation: Tools and Techniques for Determining the Value of Any Asset. 3rd ed. Hoboken: John Wiley & Sons; 2012. p. 183.
- 118. Damodaran A. Valuing Private Firms. Investment Valuation: Tools and Techniques for Determining the Value of Any Asset. 3rd ed. Hoboken: John Wiley & Sons; 2012. p. 667–701.
- 119. Anon. Daily Treasury Curve Yield Rates [Internet]. U.S. Department of the Treasury Resource Center. 2013 [cited 2013 Dec 6]. Available from: http://www.treasury.gov/resource-center/data-chart-center/interestrates/Pages/TextView.aspx?data=yieldYear&year=2013
- 120. Thilakaratne R, Brown T, Li Y, Hu G, Brown R. Mild catalytic pyrolysis of biomass for production of transportation fuels: a techno-economic analysis. Green Chem. The Royal Society of Chemistry; 2014 Sep 18;
- 121. Dutta A, Dowe N, Ibsen KN, Schell DJ, Aden A. An economic comparison of different fermentation configurations to convert corn stover to ethanol using Z. mobilis and Saccharomyces. Biotechnol Prog. Wiley Subscription Services, Inc., A Wiley Company; 2010;26(1):64–72.

- 122. Tao L, He X, Tan ECD, Zhang M, Aden A. Comparative techno-economic analysis and reviews of n-butanol production from corn grain and corn stover. Biofuels, Bioprod Biorefining. 2013 Dec 12;n/a–n/a.
- 123. Klein-Marcuschamer D, Simmons BA, Blanch HW. Techno-economic analysis of a lignocellulosic ethanol biorefinery with ionic liquid pre-treatment. Biofuels, Bioprod Biorefining. John Wiley & Sons, Ltd.; 2011;5(5):562–9.
- 124. Zhu Y, Jones SB, Biddy MJ, Dagle RA, Palo DR. Single-step syngas-to-distillates (S2D) process based on biomass-derived syngas – A techno-economic analysis. Bioresour Technol. 2012;117(0):341–51.
- 125. Anon. Ethanol Facilities Capacity by State and Plant [Internet]. Official Nebraska Government Website. 2013 [cited 2013 Jan 9]. Available from: http://www.neo.ne.gov/statshtml/122.htm
- 126. Lane J. Gusher! KiOR starts production of US cellulosic biofuels at scale [Internet]. Biofuels Digest. Biofuels Digest; 2012 [cited 2014 Apr 1]. Available from: http://www.biofuelsdigest.com/bdigest/2012/11/09/gusher-kior-starts-production-ofus-cellulosic-biofuels-at-scale/
- 127. Anon. About Dynamic Fuels [Internet]. 2013 [cited 2013 Jan 9]. Available from: http://www.dynamicfuelsllc.com/about.aspx
- 128. Anon. FutureFuel Biofuels [Internet]. 2013 [cited 2013 Jan 9]. Available from: http://www.futurefuelcorporation.com/biobased.html
- 129. Esty BC. Improved Techniques for Valuing Large-Scale Projects. J Struct Financ. Institutional Investor Journals; 1999 Jan 5;5(1):9–25.
- 130. Anon. President Bush's State of the Union Address. The Washington Post. Washington, DC; 2006;
- 131. Maron DF. Much-Touted Cellulosic Ethanol Is Late in Making Mandated Appearance. The New York Times. Business D. 2011;
- 132. American Petroleum Institute v. Environmental Protection Agency [Internet]. 2013
 [cited 2013 Aug 18]. Available from: http://www.cadc.uscourts.gov/internet/opinions.nsf/A57AB46B228054BD85257AFE
 00556B45/\$file/12-1139-1417101.pdf
- 133. Wald M. A Fine for Not Using a Biofuel That Doesn't Exist. The New York Times. Late. Washington, DC; 2012;B1.
- 134. Dutta A, Talmadge M, Hensley J, Worley M, Dudgeon D, Barton D, et al. Technoeconomics for conversion of lignocellulosic biomass to ethanol by indirect

gasification and mixed alcohol synthesis. Environ Prog Sustain Energy. John Wiley & Sons, Inc.; 2012;31(2):182–90.

- 135. Parker N. Spatially Explicit Projection of Biofuel Supply for Meeting Renewable Fuel Standard [Internet]. Transportation Research Board. 2012 [cited 2014 Jan 31]. p. 72–9. Available from: http://trb.metapress.com/content/m15q225434558612/
- 136. Anon. Unified Facilities Criteria (UFC) DoD Facilities Pricing Guide for FY 2011. Washington D.C.; 2013.
- 137. Petter R, Tyner WE. Technoeconomic and policy analysis for corn stover biofuels. ISRN Econ. 2014;2014.
- 138. Verser D. ZeaChem [Internet]. Pacific Biomass. Washington Bioenergy Research Symposium; 2010 [cited 2014 Apr 1]. Available from: http://www.pacificbiomass.org/documents/Verser %28D1%29.pdf
- Meyer PA, Tews IJ, Magnuson JK, Karagiosis SA, Jones SB. Techno-economic analysis of corn stover fungal fermentation to ethanol. Appl Energy. 2013;111(0):657–68.
- 140. Elliott DC, Hart TR, Neuenschwander GG, Rotness LJ, Zacher AH. Catalytic hydroprocessing of biomass fast pyrolysis bio-oil to produce hydrocarbon products. Environ Prog Sustain Energy. John Wiley & Sons, Inc.; 2009;28(3):441–9.
- 141. Anon. Richardson International Construction Factors Manual. Pahrump; 2007.
- 142. Anon. ENR Cost Indexes in 20 Cities 1978-2012 [Internet]. ENR Construction Economics. 2013 [cited 2014 Apr 8]. Available from: http://enr.construction.com/economics/
- 143. Anon. Range of State Corporate Income Tax Rates [Internet]. Federation of Tax Administrators. 2014 [cited 2014 Mar 16]. Available from: http://www.taxadmin.org/fta/rate/corp_inc.pdf
- 144. Anon. Weekly Retail Gasoline and Diesel Prices [Internet]. Energy Information Administration. Energy Information Administration; 2013 [cited 2013 Mar 20]. Available from: http://www.eia.gov/dnav/pet/pet_pri_gnd_dcus_nus_a.htm
- 145. Brown TR, Wright MM. A framework for defining the economic feasibility of cellulosic biofuel pathways. Biofuels (under Rev. 2014;
- 146. Anon. U.S. Natural Gas Prices (Industrial) [Internet]. Administration USEI, editor. Energy Information Administration. Washington, DC: Department of Energy; 2014 [cited 2014 Jan 30]. Available from: http://www.eia.gov/dnav/ng/ng_pri_sum_a_EPG0_PIN_DMcf_a.htm

- 147. EIA. Gasoline and Diesel Fuel Update [Internet]. Energy Information Administration; 2014 [cited 2014 Apr 2]. Available from: http://www.eia.gov/oil_gas/petroleum/data_publications/wrgp/mogas_history.html
- 148. EIA. U.S. Propane Sales by Refiners [Internet]. Washington, D.C.: Department of Energy; 2014 [cited 2014 Apr 1]. Available from: http://www.eia.gov/dnav/pet/hist/LeafHandler.ashx?n=PET&s=EMA_EPLLPA_PTG _NUS_DPG&f=M
- 149. Anon. Louisiana Quarterly Report of Forest Products [Internet]. Louisiana Department of Agriculture & Forestry; 2014 [cited 2014 Mar 1]. Available from: http://www.ldaf.state.la.us/portal/Offices/Forestry/ForestryReports/QuarterlyReportof ForestProducts/OriginalReports/tabid/452/Default.aspx
- 150. Spearman C. The proof and measurement of association between two things. Am J Psychol. 1904;15(1):72–101.